AN INVESTIGATION OF THE IMPACT OF SUBLINGUAL IMMUNOTHERAPY IN EXPERIMENTAL MODELS OF FOOD ALLERGY AND ANAPHYLAXIS

AN INVESTIGATION OF THE IMPACT OF SUBLINGUAL IMMUNOTHERAPY IN EXPERIMENTAL MODELS OF FOOD ALLERGY AND ANAPHYLAXIS

By SIYON GADKAR, BHSc.

A thesis submitted to the School of Graduate Studies in partial fulfillment of the requirements for the degree Master of Science

McMaster University © Copyright by Siyon Gadkar, September 2022

MASTER IN MEDICAL SCIENCES (2022)	McMaster University
Medical Science – Infection & Immunity	Hamilton, ON

TITLE:An investigation of the impact of sublingual immunotherapy in
experimental models of food allergy and anaphylaxis

AUTHOR: Siyon Gadkar, BHSc. (McMaster University)

SUPERVISOR: Dr. Manel Jordana

NUMBER OF PAGES: xii, 129

Lay Abstract

Food allergy is a potentially life-threatening disease which is primarily mediated by IgE antibodies. Strict allergen avoidance and use of rescue epinephrine upon accidental allergen exposure remain the standard of care. Oral immunotherapy, where individuals ingest small amounts of allergen, is currently the experimental treatment of reference to induce clinical tolerance; however, it is accompanied by a significant rate of adverse reactions. In contrast, sublingual immunotherapy (SLIT), which is less efficacious, upholds a superior safety profile. The primary objective of this thesis was to investigate the impact of SLIT in inducing clinical and immunological changes in murine models of food allergy. We demonstrated that when administered prophylactically, SLIT prevents mice from undergoing anaphylaxis. When administered to sensitized mice in a pre-allergic state, SLIT was protective against severe clinical reactivity after challenge. In conclusion, the work presented here establishes a useful platform to investigate the mechanisms underlying SLIT-mediated protection.

Abstract

Food allergy is a potentially life-threatening disease affecting up to 10% of individuals in Western countries. Clinical reactivity to food allergens is primarily mediated by immunoglobulin (Ig) E, with symptoms ranging from mild urticaria to anaphylaxis. Currently, food allergy remains a disease without a cure. Oral immunotherapy (OIT), which involves consuming small amounts of allergen, remains an experimental treatment in Canada, although has been approved by the Food and Drug Administration (FDA) in the United States for treatment of peanut allergy. While efficacious to induce desensitization, OIT is accompanied by a significant rate of adverse effects. Sublingual immunotherapy (SLIT) is a novel route of treatment for food allergy, where small amounts of allergen are placed under the tongue and held for 2-3 minutes. In contrast to OIT, SLIT offers not only treatment efficacy but also promises an excellent safety profile.

The first objective of this thesis was to first develop a SLIT regimen in murine models of food allergy where sensitization is carried out either epicutaneously or intragastrically. Secondly, we investigated the efficacy of SLIT in modulating the clinical and humoral responses in prophylactic and semi-therapeutic settings. In the prophylactic setting, where SLIT was administered prior to sensitizing allergen exposures, SLIT-treated mice were completely protected from allergic sensitization including absent production of serum ovalbumin-specific IgE. In the semi-therapeutic setting, where SLIT was administered to mice primed to develop food allergy, it produced a partial protection against food-induced clinical reactivity. This was associated with lower levels of IgE production in comparison to non-treated, allergic mice. Together, this work provides both an optimized SLIT

protocol, as well as evidence on the efficacy of SLIT in the treatment of food allergy in murine models. These findings will aid future work investigating the cellular and molecular mechanisms underlying SLIT-induced protection.

Acknowledgements

I would like to start with a tremendous thank you to my supervisor, Manel, for his patience and guidance throughout my journey at the Jordana/Waserman Lab. From my first entry into the field of research, to now graduating with my Masters, this journey has taught me more than any course I could have taken. I would also like to take this opportunity to thank my committee members, Dr. Susan Waserman, and Dr. Rodrigo Jiménez-Saiz. Thank you both for your guidance and mentorship throughout my graduate career.

A special thanks to everyone at ALK-Abello for granting us with the opportunity to start and take lead with this project. Particularly I would like to say thank you to Dr. Peter Sejer Andersen and the late Jens Brimnes. You both have been instrumental in the success of this project. Thank you for the great ideas and suggestions that were crucial for optimization of the SLIT model. Jens, you will be greatly missed.

Within my time at the Jordana/Waserman Lab, I had the pleasure of working alongside many outstanding individuals whom I now consider friends. I would like to say thank you to all the past and current members of the lab. You all have played a critical role in my journey here, and I am truly thankful that we met. Tina Walker, thank you for everything you have done. From training me when I first joined, to carrying out 100+ sample ELISAs. I truly appreciate all the help you have provided me. Thank you to Jianping Wen for teaching me and supporting me throughout my journey. Dr. Roopali Chaudhary, thank you for welcoming me into the lab, mentoring me, and for always lending an ear when I needed help. Dr. Joshua Koenig, I will never forget all that we have done together and how you have helped me grow. I will remember our times not just working together, but as friends outside the lab. To past and current lab members, Malcolm Davidson, Saba Manzoor, Owen Baribeau, Emily Grydziuszko, Sharon Khavkine-Binstock, Atai Ariaz, Emily Grydziuszko, Olivia Mann-Delany, thank you all for assisting me and supporting me during my time at the Jordana Lab. Lastly, to Allyssa Phelps and Dr. Kelly Bruton, you both have made this experience a joyous one, and I will never forget the amazing memories we made. Although I am leaving, I hope to keep in touch and make some more memories.

Dr. Paul Spill, we have been through quite some adventures, and I am glad that you made my experience at the lab that much better. Thanks for pushing me both mentally and physically (at the gym) to really test myself. Light weight baby. To Sonya, who has provided her unconditional love & support no matter how annoying I got. Thank you for sticking by me, I really do appreciate it. Lastly, to my parents, who have provided all the support I could have ever asked for, and much more. I hope that I can continue to make you both proud.

Table of Contents

Confirmation of the protection against epicutaneous sensitization and subsequent
clinical reactivity by 9-weeks of semi-therapeutic OVA-SLIT
Evaluation of semi-therapeutic OVA-SLIT in establishing protection against
intragastric sensitization71
Lowering the dose of OVA-SLIT confers partial protection from food-induced clinical
reactivity within intragastric sensitized mice76
Chapter 4: Discussion
Prophylactic OVA-SLIT is protective against both epicutaneous and intragastric
allergic sensitization, and subsequent food-mediated anaphylaxis
Semi-therapeutic OVA-SLIT provides protection against food-mediated anaphylaxis in
both the epicutaneous and intragastric models of allergic sensitization
Semi-therapeutic OVA-SLIT in the epicutaneous sensitization model
Semi-therapeutic OVA-SLIT in the intragastric sensitization model
Summary of data findings and future directions
Tables 94
Appendices
References

List of Figures & Tables

Figure 1. Schematics of models of allergic sensitization40
Figure 2. Visual representation of epicutaneous sensitization40
Figure 3. Visual representation of the SLIT administration protocol41
Figure 4. Prophylactic SLIT protects mice against epicutaneous allergic sensitization and
allergen-induced clinical reactivity42
Figure 5. Prophylactic SLIT prevents the emergence of OVA-specific IgE after
epicutaneous sensitization
Figure 6. Prophylactic SLIT protects mice against intragastric sensitization and food-
induced clinical reactivity
Figure 7. Prophylactic SLIT prevents the emergence of OVA-specific IgE after intragastric
sensitization
Figure 8. Semi-therapeutic SLIT (6/9 wk) partially protects epicutaneous sensitized mice
from food-induced clinical reactivity56
Figure 9. Semi-therapeutic SLIT partially prevents the emergence of OVA-specific IgE
after epicutaneous sensitization
Figure 10. Impact of semi-therapeutic SLIT (12/15 wk) on food-induced clinical reactivity
in epicutaneously sensitized mice
Figure 11. Impact of semi-therapeutic SLIT on the emergence of OVA-specific IgE and
IgG1 after epicutaneous sensitization
Figure 12. Experimental repeat of the 9-week semi-therapeutic SLIT protocol confirms
protection from food-induced clinical reactivity in epicutaneously sensitized mice67

Figure 13. Impact of 9-week SLIT treatment on the emergence of OVA-specific IgE in
epicutaneously sensitized mice
Figure 14. Semi-therapeutic SLIT treatment for 9 weeks was not protective against
intragastric sensitization and subsequent food-induced clinical reactivity71
Figure 15. Semi-therapeutic SLIT does not prevent the emergence of OVA-specific IgE
and IgG1 antibodies after intragastric sensitization74
Figure 16. Lower dosed semi-therapeutic SLIT partially protects intragastric sensitized
mice from food-induced clinical reactivity76
Figure 17. Semi-therapeutic SLIT partially prevents the emergence of OVA-specific IgE
after intragastric sensitization
Table 1. Clinical grading scheme for evaluating mice undergoing a systemic allergen
challenge94

Abbreviations

α-Gal	Galactose-a-1,3-galactose
ANOVA	Analysis of Variance
BC	B Cell
Breg	B Regulatory Cell
CHILD	Canadian Health Infant Longitudinal Development
СТ	Cholera Toxin
DBPC	Double-Blinded, Placebo-Controlled
ELISA	Enzyme-Linked Immunosorbent Assay
EPIT	Epicutaneous immunotherapy
FDA	U.S. Food and Drug Administration
GI	Gastrointestinal
Ig	Immunoglobulin
i.g.	Intragastric
IFN	Interferon
MBC	Memory B Cells
ODN	Oligodeoxynucleotides
OFC	Oral Food Challenge
OIT	Oral Immunotherapy
OVA	Ovalbumin
PBS	Phosphate Buffered Saline
PC	Plasma Cell

PN	Peanut
s.c.	Subcutaneous
SCIT	Subcutaneous Immunotherapy
SEM	Standard Error of Mean
SLIT	Sublingual Immunotherapy
SPINK5	Serine Peptidase Inhibitor Kazal Type 5
SPT	Skin Prick Test
ТС	T cells
Tfh	T Follicular Helper
Treg	Regulatory T Cell

Chapter 1: Background

1.1 Food Allergy

Food allergy is characterized by a pathogenic immune response against innocuous food antigens. Currently, challenge-diagnosed food allergy affects up to 10% of individuals in Western countries ^{1–4}. In Canada, the self-reported prevalence of food allergy is estimated at 6.67%, with peanut (PN) allergy affecting approximately 1.5% of individuals ⁵. However, another study found that physicianreported food allergy occurs in 2.5% of Canadians, significantly lower than the selfreported prevalence ⁶. Reactivity against food allergens ranges in severity as patients may experience reactions such as urticaria (skin rash), abdominal cramps, diarrhea, or coughing/trouble breathing, that are often restricted to a localized site ⁷. In severe cases, patients may experience a systemic response known as anaphylaxis. This response affects multiple organs and can be life-threatening if not treated immediately with rescue medication, such as epinephrine ⁷.

Some allergies, such as those to milk and egg are often outgrown. ¹. For example, egg allergy is commonly acquired during the first year of life ⁸, and one retrospective study of 881 patients found that the median age of outgrowing egg allergy was 9 years old ⁹. Similarly, milk allergy also presents during the first year of life, with the median age for outgrowing being 10 years of age ^{8,10}. In contrast, allergies to tree nuts, PN, fish or shellfish are generally lifelong with less than 20% of patients outgrowing their allergy by 20 years of age ¹⁰. Despite the persistence of these allergies, no curative treatments exist, and the current standard of care is

strict allergen avoidance. However, cases of accidental exposures occur with an incidence rate of 14.3% annually, especially on cultural holidays, such as Easter and Halloween, when treats are common ^{11,12}. Food allergies further present a significant impact on the quality of life (QoL) of allergic individuals and their families ^{13–16}. QoL questionnaires administered to allergic individuals consistently report a worse QoL in comparison to individuals type 1 diabetes mellitus or rheumatological diseases ^{13–16}. Additionally, family members and or caregivers of the allergic individual report a similar impairment in QoL ¹³. This is largely attributed to the possibility of undergoing anaphylaxis upon accidental exposures. In combination with the need to maintain allergen-free safe environments, this often increases anxiety amongst the allergic patients and their families ^{13–16}. Overall, the economic cost for food allergy was estimated at USD24.8 billion annually in the United States, indicating a substantial burden for both the healthcare system and the families of allergic individuals ¹⁷.

1.2 Mechanisms of food-induced reactivity

Allergic reactions to innocuous food allergens are a type 1 hypersensitivity reaction that is primarily mediated by IgE⁷. Typically associated with TH2 immunity, allergen-specific IgE has a short half-life of 2 days in circulation in humans ¹⁸. However, evidence in mice has shown that IgE can remain bound to its high-affinity receptor (FccR1) on the surface of mast cells for up to 67.3 days (CI 46.5-88.1 days) ¹⁹. Upon a secondary allergen exposure, food allergens can cross-link multiple FccR1-bound IgE antibodies to induce the degranulation of mast cells

and basophils thus leading to the release of preformed granules as well as rapidly synthesized mediators, such as PAF ⁷. These granules contain vasoactive mediators, such as histamine, which are responsible for the signs and symptoms of an allergic reaction 20,21 . Histamine for example, has been shown to bind receptors on endothelial cells to cause vasodilation and a decrease in blood pressure, whereas PAF seems to be responsible for the magnitude and duration of the anaphylactic response 20,21 . During the degranulation phase, other mediators such as heparin, serotonin, proteases, tryptases, chymases, prostaglandins and leukotrienes contribute to the anaphylactic reaction $^{20-22}$. Together, they mediate a wide array of physiological effects including airway constriction, increased mucous production, itching, and diarrhea, amongst others $^{20-22}$.

Following the immediate allergic response, patients may experience a secondary reaction known as a late phase response ^{22,23}. This response is primarily mediated by T helper 2 (TH2) cells through the secretion of interleukins (IL) such as IL-4, IL-5, and IL-13. In combination with mediators released during the acute phase of the allergic reaction, TH2 cells can recruit various immune cells such as eosinophils and basophils, in turn causing a secondary rise in histamine after the initial allergic reaction ^{22,23}. The secondary late-phase response is typically not lethal.

1.3 Skin as a potential site of allergic sensitization

Traditionally, the gut was considered the primary route for allergic sensitization because the gastrointestinal (GI) tract is the main site of interaction with food

allergens. However, individuals typically present to the hospital upon their first known allergen exposure, which suggests a prior allergic sensitization event that is clinically silent ^{8,24}. Numerous factors have been identified as contributors to this silent allergic sensitization. It has been shown that an altered gut microbiome during the early years of life is associated with food sensitization ²⁵. The Canadian Health Infant Longitudinal Development (CHILD) study found that a low microbiota richness in combination with elevated enterobacteriacaea/bacteroidaceae ratio by 3 months of age is associated with early food sensitization ²⁵. Therefore, these data identify a critical period of immunological development where environmental factors may facilitate allergic sensitization. However, the immune system remains largely tolerant in early life, thus providing an opportunity for interventions to prevent allergic sensitization. One study assigned 640 infants with severe eczema and/or egg allergy to either consume or avoid ingestion of PN until 60 months of age ²⁶. Amongst this at-risk population, the prevalence of PN allergy at 60 months was 13.7% in the avoidance group and only 1.9% in the consumption groups for those with an initial negative skin-prick test. For individuals with an initial positive egg allergy skin-prick test, the avoidance group had a prevalence of 35.3% and the consumption group of 10.6% 26 . The study demonstrates that early oral introduction of PN dramatically decreases the frequency of food allergy amongst high-risk populations.

Allergen exposures can occur at any one of the physical barriers within the body. It has become increasingly clear that the skin is another key barrier through

4

which individuals may become sensitized against food allergens. The skin behaves differently than the GI tract, as the outmost epidermal layer serves as a barrier against the entry of any antigens. Although barrier disruption through injury may occur, the rapid release of inflammatory cytokines by structural cells and activation of resident immune cells prevents the onset of any infection. In 2003, a study examined a cohort of children with a history of PN allergy, confirmed by a doubleblind PN challenge ²⁷. Analysis of cord blood showed no PN-specific IgE, indicating a lack of prenatal sensitization; however, a significant relationship was found between the onset of PN allergy and the use of PN oil for treatment of inflamed skin²⁷. Approximately 91% of individuals, in comparison to 53% of atopic controls and 59% of normal controls, had been exposed to skin creams containing PN oils ²⁷. Therefore, it was proposed that exposure to PN proteins on inflamed skin could be an avenue through which allergic sensitization occurs. As the PN protein is largely innocuous, it is likely that alarmins released as a result of inflamed skin are responsible for this phenomenon.

Additional studies have found associations between the filaggrin gene and missense mutations in the serine peptidase inhibitor Kazal type 5 (SPINK5) gene and the risk for IgE-mediated food allergy ^{28,29}. Both genes are responsible for maintaining the integrity of the epithelial barrier. Therefore, it was hypothesized that a disrupted epithelial barrier could lead to further complications such as atopic dermatitis and allergic sensitization ^{28,29}. Genotyping of allergic and non-allergic individuals revealed that a loss-of-function in the filaggrin gene and missense

mutations in the SPINK5 gene are significantly associated with a risk of IgEmediated food allergy ^{28,29}. In addition to genetic mutations, diseases such as atopic dermatitis are also associated with an increased risk of PN allergy. Ha *et al* found a dose-response relationship between the levels of PN-protein in household dust and PN-skin prick test results amongst patients with atopic dermatitis, with a higher PNprotein concentration attributing to an increase in severity of the skin prick test ³⁰. Together, these studies provide evidence to illustrate sensitization to food allergens through a disrupted epithelial barrier.

1.4 Persistence of allergy and the IgE recall response

It was traditionally thought that long-lived IgE⁺ plasma cells (PC) were the reservoir of allergic humoral memory. PCs are terminally differentiated B cells (BC) that reside primarily in the bone marrow and are responsible for the production of antibodies. As seen in the anti-viral IgG response, the antibody half-life ranges from 50 years for varicella-zoster virus, to more than 200 years for measles and mumps ³¹. Additionally, in a murine model of house dust mite allergy, repeated chronic allergen exposure for 15 months demonstrated the emergence of an IgE⁺ PC population, capable of inducing anaphylaxis ³². However, this notion has been inconsistent with observations in both human and murine studies of food allergy. Multiple studies have concluded that IgE titres often wean with allergen avoidance. For example, patients allergic to pollen demonstrate an increase in serum IgE during pollen season, which wean off with the conclusion of the pollen season ³³.

avoidance for at least 6-10 months resulted in a drastic decrease of both total and allergen-specific IgE titres ³⁴. When examining the entirety of the patient-specific follow-up periods, a median of 76% decrease in allergen-specific IgE titres was noted ³⁴. However, upon consumption of the fish allergen, IgE titres and clinical symptoms re-emerged ³⁴. This finding is consistent with a recent report of the decline in galactose- α -1,3-galactose (α -Gal)-specific IgE titres in patients who avoid tick bites 35 . α -Gal is an oligosaccharide of non-primate mammals, that is targeted by α -Gal-IgE within individuals allergic to red meat ^{35,36}. Interestingly, subjects with serum IgE antibodies against α -Gal often have a history of tick bites ³⁶. Therefore, an association was established between IgE sensitization against α -Gal, and a history of tick bites. Together these observations suggest that despite the decline in IgE antibodies with allergen avoidance, the capacity to regenerate IgE antibodies and subsequent clinical reactivity are long-lived. In a murine model of PN allergy, our lab demonstrated that allergen-specific IgE titres and IgE⁺ PCs contract and are undetectable 3-6 months post-allergic sensitization, although, the capacity to regenerate IgE antibodies persisted for over 15 months post-allergic sensitization¹⁹. These findings suggest a role for memory (M) BCs in reconstituting the IgE⁺ PC population and the persistence of IgE-mediated food allergy.

Upon antigen exposure, MBCs undergo rapid proliferation and differentiation into PCs ³⁷. In human food allergy, IgE⁺ MBCs were thought to be the reservoir of IgE-secreting cells. However, it has been demonstrated that IgE⁺ MBCs in humans are extremely rare, and perhaps even absent ^{38,39}. These findings

were derived from human peripheral blood mononuclear cells (PBMCs), so they do not exclude the possibility of tissue resident IgE⁺ MBCs. Interestingly, current literature points to a subset of non-IgE⁺ MBCs which contribute to long-lived IgE immunity. For context, IgE antibodies can derive from two distinct pathways: 1) a direct pathway, where a naïve IgM⁺ BC undergoes class switch recombination directly to IgE, or 2) a sequential pathway, where BCs first switch to an intermediate subclass prior to class switching to IgE⁴⁰. Analysis of IGH repertoires in human PBMCs identified clonal lineages of IgE BCs derived predominantly from IgG and partially from IgA-expressing BCs⁴⁰. These findings are consistent in murine models of allergy $^{41-44}$. These studies suggest that IgE⁺ PCs primarily originate from a memory population of IgG1⁺ BCs. In this regard, our lab demonstrated that IgG1⁺ MBCs persist even after the decline of serum IgE and loss of clinical reactivity ¹⁹. Furthermore, non-sensitizing allergen exposure up to 15 months post-sensitization generated IgE antibodies in serum ¹⁹. In addition, in a helminth model of TH2 immunity, IgG1⁺ MBCs and IL-4⁺ CD4⁺ memory T cells isolated from mice infected with Nippostrongylus brasiliensis (N.b.) could induce a N.b.-specific IgE response upon re-infection when adoptively transferred to Rag1⁻ $^{-}$ recipient mice 42 . Since the recipient mice lack the ability to generate lymphocytes, it can then be concluded that the IgE response was generated from the transferred IgG1⁺ MBCs. Collectively, these studies spotlight long-lived IgG1⁺ MBCs as the reservoir of the IgE allergic response in humans and mice.

Activation of IgG1⁺ MBCs can often occur through accidental allergen exposures, which in turn, drive the emergence of IgE-secreting cells and thus the maintenance of food-specific IgE titres. This secondary response, the allergic recall response, is strictly dependent on T cells (TCs)^{45,46}. Without the presence of CD4⁺ TCs, the production of IgE antibodies and subsequent clinical reactivity does not occur ^{45,46}. It is important to note that naïve CD4⁺ TCs, rather than antigen experienced CD4⁺ TCs, are sufficient to propagate the recall response ⁴⁵. In vitro models suggest an avenue through how CD4⁺ TCs help during the secondary allergic response. Here, PN allergic mice were first treated with anti-CD4 antibody or vehicle in vivo to deplete CD4⁺ TCs¹⁹. Next, splenocytes were harvested, labelled with CFSE, and stimulated ex vivo with PN in the presence of blocking antibodies against IL-4, CD40 ligand (CD40L) and/or an isotype control ¹⁹. In the context of MBCs, proliferation upon allergen stimulation was observed in a CD4⁺ TC-dependent, IL-4- and CD40L-independent manner¹⁹. In contrast, proliferation of plasmablasts was impaired by the lack of both CD4⁺ TCs and IL-4¹⁹. Therefore, this data suggests the importance of CD4⁺ TCs in the differentiation of MBCs into plasmablasts, dependent on IL-4 secretion.

IL-4 is a critical signalling molecule to initiate IgE class switch recombination ^{47,48}. IL-4 interacts with one of two heterodimeric receptors, type I and type II. The type I receptor incorporates the IL-4R α chain with the common γ chain, whereas the type II receptor combines the IL-4R α chain with the IL-13R α 1 chain ^{49,50}. With respect to the source of IL-4, studies utilizing genetically engineered IL-4 reporter mice, found that TH2 cells within the lymph nodes produced both IL-4 and IL-13, whereas T follicular helper cells (Tfh) primarily produced IL-4 in response to parasitic helminth infection ⁵¹. The type II receptor has a secondary ligand, known as IL-13. In murine models of asthma, neutralization of IL-13 attenuated the asthma phenotype (airway hyperresponsiveness, eosinophil recruitment, and mucus overproduction) ⁵². Similarly in atopic dermatitis, Bitton *et al.* demonstrated that dermatitis symptoms and expression of TNF- α (inflammatory cytokine) are dependent on IL-13 signaling via the type II receptor ⁵³.

Since both IL-4 and IL-13 share the signaling pathway through the IL-4R α chain, antibody-mediated blockade of this receptor prevents the effects of these cytokines. The efficacy of dupilumab, an anti-IL-4R α monoclonal antibody, has been studied in humans with atopic diseases such as atopic dermatitis and uncontrolled asthma ^{54,55}. For individuals with moderate-to-severe asthma, receiving subcutaneous (s.c.) dupilumab every 2 weeks for 52 weeks led to a 47.7% lower rate of severe asthma exacerbations in comparison to placebo ⁵⁴. Additionally, patients exhibited an increase in the forced expiratory volume in 1 second, indicating improvement in lung function and asthma control ⁵⁴. For individuals with moderate-to-severe atopic dermatitis, dupilumab was evaluated as a monotherapy and in combination with topical glucocorticoids ⁵⁵. With 12-weeks of dupilumab monotherapy, 85% of patients exhibited a 50% reduction in the Eczema Area and Severity Index (EASI) score in comparison to 35% of those in the placebo group ⁵⁵. Reduction in the pruritus scores, itching, of 55.7% in the

dupilumab group in comparison to 15.1% in the placebo group ⁵⁵. Combination therapy found patients using 50% less of topical glucocorticoids and a decrease in the frequency of nasopharyngitis and headache when compared to patients without dupilumab ⁵⁵. In both studies, patients receiving dupilumab demonstrated a significant reduction in disease severity and improved symptom scores. Further studies have shown that antibody-mediated blockade of IL-4 signalling through the IL-4R α chain or α -IL4 antibodies abolishes the food allergic recall response in human PBMCS and the allergic response towards helminth infections in mice ^{47,56}. Recent work from our lab has demonstrated that blockade of IL-4/IL-13 signaling with anti-IL-4R α prevented the generation of IgE antibodies in a human *in vitro* culture system, and single-cell RNA-sequencing analysis demonstrated a skewed T cell polarization, away from the TH2 phenotype ⁵⁶. Furthermore, treatment with anti-IL-4R α abolished the generation of IgE in allergic mice and prevented anaphylaxis upon a systemic allergen challenge ⁵⁶. Importantly, anti-IL-4R α treated allergic mice remained unresponsive to a PN challenge 6 weeks after discontinuation of anti-IL-4R α treatment ⁵⁶. Therefore, anti-IL-4R α may potentially induce extended protection and, perhaps some form of immunological reprogramming. These findings suggest a vital role for anti-IL-4Ra treatment in the future of allergen immunotherapy.

1.5 Current state of allergen immunotherapy

Currently, no curative treatments exist for food allergy and, thus, the standard of care remains strict allergen avoidance ⁵⁷. Epinephrine and

antihistamines relieve allergic symptoms only after a reaction has occurred. Various allergen immunotherapy strategies have been extensively investigated towards establishing an efficacious treatment or even a cure for food allergy. The four principal approaches include subcutaneous immunotherapy (SCIT), epicutaneous immunotherapy (EPIT), oral immunotherapy (OIT), and sublingual immunotherapy (SLIT). Although these therapies differ in the route of allergen administration, they typically involve a similar approach to treatment ⁵⁸. An oral food challenge (OFC) determines the initial starting dose of allergen. After 1-2 weeks, patients begin small incremental dose escalations for weeks to months, until a predetermined maintenance dose is reached. Maintenance doses can range from low, such as 3-7 mg of allergen per day in SLIT, to high, such as 300-4,000 mg of allergen per day in OIT ⁵⁸. As reference, one PN typically contains around 300 mg of allergen protein. This maintenance dose can be extended indefinitely. Depending on the allergen immunotherapy protocol, patients undergo a second OFC after a predetermined period of the maintenance phase. A pass, defined as the ability to tolerate an increased amount of allergen in comparison to baseline measurements would indicate allergen desensitization. Serum measurements from blood collected during this second challenge often indicate a decrease in allergen-specific IgE, a decrease in basophil activation levels, and an increase in allergen-specific IgG4. Broadly, these changes suggest a cellular shift away from the allergic phenotype 58 .

1.6 SCIT

SCIT involves the injection of allergen under the skin of the arm (subcutaneously) and has been shown to be efficacious in the treatment of seasonal allergic rhinitis ^{59,60}. A Cochrane meta-analyses on the efficacy of SCIT for allergic rhinitis demonstrated a standardized mean difference of -0.73 for symptom scores, favouring SCIT over placebo⁵⁹. This was accompanied by a reduction in medication score (use of medication for relief of allergic symptoms) of -0.57, favouring SCIT over placebo⁵⁹. Double-blind, placebo-controlled (DBPC) clinical trials of SCIT for ragweed and grass pollen demonstrated an approximate 30-40% reduction in both symptom and medication score during 3-4 years of immunotherapy ^{61–63}. This improvement was maintained for up to 3 years after discontinuation of immunotherapy; however, allergen reactivity did return thereafter ^{59,61–63}. A recent phase 2a, multicenter, DBPC parallel group study of adults with grass pollen-induced seasonal allergic rhinitis examined the effects of dupilumab in combination with SCIT⁶⁴. The study found that 61.5% of patients in the SCIT + dupilumab group reached maintenance dose, versus 46.2% of patients who received SCIT alone 64 . Additionally, only 7.7% of SCIT + dupilumab-treated patients required rescue medication (epinephrine) as compared to 19.2% of patients receiving SCIT alone ⁶⁴. Withdrawal rate was also reported to be significantly lower in the combinational therapy (7.7%) as compared to SCIT alone $(30.8\%)^{-64}$. However, this study found no significant differences in total nasal symptom scores between both treatments ⁶⁴. Although dupilumab treatment did not achieve a better

reduction in symptom scores, it did increase patient adherence, which is critical in the treatment of food allergy.

SCIT can result in adverse systemic effects, and thus, often confines allergen administration to a clinical setting with medical supervision, and access to rescue medication, such as epinephrine ^{59,60}. The frequency of adverse events is likely the reason for the discontinuation of clinical trials involving the treatment of food allergy with SCIT. One study reported a 67-100% decrease in symptoms (nausea, vomiting, urticaria, rhinitis, mucosal pruritus, asthma, and hypotension) for PN-allergic patients receiving PN-SCIT compared to baseline measurements; however, this was accompanied by a 13.3% rate of systemic reactions ⁶⁵. A second study examining the effects of SCIT for PN allergy reported a 23% systemic reaction rate during the escalation phase and 39% during the maintenance phase of immunotherapy, often treated with epinephrine ⁶⁶. Ultimately, these findings discouraged the use of SCIT for the treatment of food allergy.

1.7 EPIT

EPIT utilized a novel route of allergen delivery, through an epidermal patch containing the allergen ^{58,67}. These patches are usually placed on the back or upper arm and changed at 24-48-hour intervals over the course of treatment ^{58,67}. This treatment was initially explored in a DBPC trial for the treatment of grass pollen allergy ⁶⁸. Patients received treatment before and during the pollen season, with the primary outcome being nasal provocation tests ⁶⁸. Patients receiving the epidermal patch demonstrated an improvement in the nasal provocation tests ⁶⁸. A pilot study

examining the safety and efficacy of cow's milk EPIT was reported in 2010⁶⁹. The study followed a 3-month, DBPC trial in 18 children (mean age of 3.82 years), with treatment consisting of three 48-hour epidermal patch application on the upper back ⁶⁹. The cumulative tolerated dose was evaluated by OFC conducted in a clinical setting ⁶⁹. Adverse reactions to the patch were localized to the area of application, consisting of local eczema or erythema⁶⁹. Although being well tolerated, the EPIT treatment failed to significantly improve the cumulative tolerated dose in the active group in comparison to placebo⁶⁹. A 2016 randomized DBPC study examined the safety and efficacy of PN-EPIT⁷⁰. Participants were treated with either 100 or 250 μ g PN, or placebo and examined for treatment success after 52 weeks ⁷⁰. The primary outcome was defined as passing a 5,044 mg protein OFC or achieving a 10-fold or greater increase in dose consumption from baseline ⁷⁰. At 52 weeks, 12% of placebo-treated participants, 46% of 100 µg PN participants, and 48% of 250 µg PN participants achieved treatment success ⁷⁰. Approximately 14.4% of placebo doses and 79.8% of EPIT doses resulted in predominantly local patch-site reactions ⁷⁰. One phase 2b DBPC, dose-ranging trial assigned patients to receive a patch containing placebo, 50, 100 or 250 µg of PN protein ⁷¹. The study was conducted for 12 months, with a DBPC food challenge to evaluate changes in eliciting doses ⁷¹. At 12 months, a statistical significance was seen only between the 250 μ g group (50% of patients) and placebo (25% of patients) ⁷¹. When stratified for age, a statistical difference was found for the 6- to 11-year-old cohort at 250 µg, but no differences were found between the adolescents/adults receiving the same dose ⁷¹. A recent phase 3, randomized, DBPC trial examined the efficacy of a patch containing 250 µg of PN protein *versus* placebo ⁷². The study had a high treatment adherence (98.5%) but found the success rate, defined as percentage of patients meeting a defined eliciting dose to PN OFC, of the treatment group to be only 35.3% in comparison to 13.6% with placebo ⁷². The pre-specified lower bounds of the confidence interval were not met, and thus, the authors could not report a positive trial result ⁷². These studies demonstrated an initial increase, followed by a decrease in PN-IgE, increases in PN-IgG4, and reduction in basophil activation and PN-specific TH2 cytokines ^{58,67,69–72}. Although safe, EPIT remains contentious regarding efficacy, but the high degree of safety holds merit for further exploration in treatment of food allergy.

1.8 OIT

OIT is currently the leading experimental therapy for the treatment of food allergy, with peanut OIT (Palforzia) having been approved by the FDA in the United States of America. It involves ingesting small, tolerable amounts of food allergens that are gradually increased to a predetermined maintenance dose that aims for approximately 300-4,000 mg of allergen ingestion per day ⁵⁸. This maintenance dose can then be continued for an indefinite amount of time. Primary outcome of clinical trials is generally desensitization, defined as the successful completion of an OFC after the maintenance phase. Individuals who can successfully tolerate the oral food challenge are deemed desensitized and those who pass an OFC after an arbitrary period of treatment cessation, have achieved

"sustained unresponsiveness" ^{58,73}. Although applicable to a variety of food allergens, OIT has predominantly been examined in the treatment of egg, milk and PN allergy ^{58,73}.

A randomized trial of egg OIT in 40 children (5-11 years) reported a 55% desensitization rate to an OFC of 5 g of egg white powder at 10 months, increasing to 75% for a 10 g egg white powder oral food challenge at 22 months ⁷⁴. Of those that were desensitized, only 27.5% of patients exhibited sustained unresponsiveness 6-8 weeks after treatment cessation ⁷⁴. Long-term treatment of patients for 4 years with egg OIT resulted in 20/40 (50%) participants achieving sustained unresponsiveness 4-6 weeks after treatment cessation ⁷⁵. Therefore, there seemed to be an association between treatment duration, and success in achieving sustained unresponsiveness, though sustained unresponsiveness still remains elusive.

Similar results were observed for cow's milk OIT. In a 2004 study, 21 cow's milk allergic children (6-10 years old) were placed on OIT or placebo ⁷⁶. Here, approximately 72% of participants achieved desensitization over the treatment course of 6 months ⁷⁶. Pajno *et al.* report a similar finding of 67% desensitization rate following 18 weeks of cow's milk OIT treatment ⁷⁷. When the treatment duration was extended to 1 year, a desensitization rate of 90% was observed in the allergic patients, although the study population included only 2–3-year-old children ⁷⁸. One study sought to examine the addition of omalizumab, an anti-IgE biologic, with cow's milk OIT ⁷⁹. This DBPC trial first randomized patients to omalizumab or placebo ⁷⁹. OIT was initiated after 4 months of omalizumab/placebo, dose

escalation for 22-40 weeks, followed by a maintenance phase. At 28 months, all patients underwent an OFC where 88.9% of omalizumab-treated subjects and 71.4% of placebo-treated subjects passed the OFC ⁷⁹. Next, OIT was discontinued for 2 months, with a rechallenge at month 32 to assess sustained unresponsiveness ⁷⁹. Here, 48.1% of patients in the omalizumab group and 35.7% of patients in the placebo group achieved sustained unresponsiveness (P = 0.18) ⁷⁹. Despite treatment with omalizumab, cow's milk OIT yields similar findings to other allergen immunotherapies in terms of the difficulties in achieving sustained unresponsiveness. Addition of omalizumab during treatment reduced the proportion of symptom-evoking doses by 14% ⁷⁹. Overall, this study demonstrates certain advantages of using a combination approach to allergen immunotherapy.

One of the first studies to explore PN OIT was reported in 2009 ^{73,80}. Subjects first had a one-day escalation phase from 0.1 mg of PN protein, doubling every 30 minutes to a final dose of 50 mg ⁸⁰. Next, there was a buildup phase where doses were increased by 25 mg every 2 weeks until 300 mg was reached ⁸⁰. The subjects then maintained 300 mg of PN protein daily for up to 36 months, after which they underwent an OFC where 300, 600, 1,200, and 1,800 mg of PN protein were given every 30 minutes for a total of 3.9 g PN-protein ⁸⁰. Approximately 93% of patients successfully ingested 3.9 g of PN protein during the OFC ⁸⁰. A decrease in the skin prick test (SPT) and activation of basophils at 6 months, and a decrease in PN-specific IgE at 12-18 months were reported ⁸⁰. On the other hand, IgG4 antibodies significantly increased ⁸⁰. Further analysis of PBMCs stimulated with

PN protein revealed an increase in the secretion of IL-10, IL-5, IFN- γ , and TNF- α at 6-12 months ⁸⁰. IL-4 and IL-17 were undetectable from baseline to after PN stimulation⁸⁰. Overall, these data suggest suppression of TH2 immunity, and perhaps an induction of a regulatory response indicated by the secretion of IL-10 and IFN-y. A shorter randomized, placebo-controlled clinical trial of PN-OIT for 48 weeks reported an 84% desensitization rate for a 5 g PN-protein OFC⁸¹. Similarly, the PN-OIT group demonstrated a decrease in skin prick test, IL-5, IL-13, and an increase in PN-specific IgG4⁸¹. Interestingly, the study further reports an increase in the ratio of FoxP3^{hi}:FoxP3^{intermediate} ratio CD4⁺CD25⁺ TCs (Treg cells) at the time of OFC in the PN-OIT treated patients⁸¹. As FoxP3 is a transcription factor associated with Treg cells, this finding suggests immune modulation provided by OIT. However, this immune modulation is temporary, as clinical reactivity returns upon treatment cessation. One clinical trial treated patients with PN-OIT for up to 5 years with a maximum dose of 4,000 mg/day of PN-protein⁸². Approximately 61.5% (24/39 subjects) of patients successfully completed the protocol ⁸². These subjects had a smaller SPT, lower levels of PNspecific IgE and lower ratios of PN-specific IgE:total-IgE in comparison with subjects who were unsuccessful in completing the study protocol⁸². Of the 24 patients, 12 (50%) achieved sustained unresponsiveness after discontinuing treatment for 1 month ⁸². Levels of PN-specific IgE, and IgG4 were no longer different between the subjects at the end of the treatment cessation period ⁸². In a recent randomized DBPC phase 2 study, participants were built up to a 4,000 mg

of PN protein per day maintenance phase lasting until week 104⁸³. Next, patients were asked either to discontinue treatment (PN-0), ingest 300 mg of PN protein daily for 52 weeks (PN-300), or receive oat flour (placebo)⁸³. OFC were conducted at weeks 104, 117, 130, 143, and 156⁸³. At the week 104 OFC, the OIT-treated groups had an approximate 84% desensitization rate, compared to 4% in the placebo group ⁸³. For the PN-0 group, the rate of sustained unresponsiveness was 54% at week 117, 20% at week 130, 15% at week 143, and 13% at week 156⁸³. The PN-300 group followed a similar trend with 54% of patients passing the OFC at week 117, 43% at week 130, 37% at week 143 and 37% at week 156⁸³. Merely 13 weeks after discontinuation of treatment, the protection gained from OIT is lost in 65% of individuals. This loss of protection does seem to be slowed down by daily consumption of a lower dose of PN protein, as at week 156, 24% more patients consuming a lower dose of PN were protected in comparison to those who discontinued treatment. Both these studies demonstrate that although OIT is efficacious in desensitizing individuals, discontinuation, or reduction in consumption of PN-protein can dramatically compromise the achieved level of protection. Therefore, patients are required to continue daily consumption of PNprotein, likely indefinitely, to maintain protection.

A major limitation of OIT lies in its safety-to-efficacy ratio. Although effective at achieving desensitization, OIT has a high rate of adverse reactions, in comparison to other forms of allergen immunotherapy, often requiring rescue medications, such as antihistamines or even epinephrine ^{58,73,84}. A recent systematic

review and meta-analysis examined randomized control trials comparing the safety of OIT for PN allergy ⁸⁴. Studies were evaluated for anaphylaxis occurrence, adverse events, epinephrine use, and quality-of-life⁸⁴. Close examination of 12 trials highlights an increase in risk of anaphylaxis (RR=3.12), an increase in anaphylaxis frequency (incidence rate ratio=2.72), and an increase in epinephrine use (RR=2.21) during OIT in comparison to no OIT⁸⁴. Additionally, OIT increased the risk of adverse events (RR=1.92), and non-anaphylactic reactions (RR=1.79), such as vomiting (RR=1.79), angioedema (2.25), upper tract respiratory reactions (RR=1.36), and lower tract respiratory reactions (RR=1.55) 84 . Some studies report up to 83% of doses resulting in a form of allergic reaction and a patient withdrawal rate up to 46% due to adverse reactions ^{73,83,85}. Additional factors, such as concurrent illness, sub-optimally controlled asthma, timing of dose administration after food ingestion, exercise and during menses can contribute to exacerbation of allergic symptoms during OIT ^{73,86,87}. Thus, patient-specific dosage adjustments may be required to prevent OIT-mediated allergic symptoms. An additional approach found that pre-treatment with omalizumab allowed all 13 patients to tolerate the 11 desensitization doses given on the first day, requiring minimal to no rescue treatment ⁸⁸. In a more recent study, patients underwent a rapid 1-day desensitization of up to 250 mg PN protein after either receiving omalizumab or placebo for 12 weeks⁸⁹. Omalizumab was then discontinued, and patients were kept on 2 g of PN protein daily⁸⁹. Omalizumab-treated patients tolerated a median PN dose of 250 mg on the initial desensitization day versus 22.5 mg for placebo-treated

subjects ⁸⁹. Additionally, 79% of omalizumab-treated patients and only 12% of placebo-treated patients passed the 4 g oral food challenge 12 weeks after cessation of omalizumab⁸⁹. Pre-treatment with omalizumab was critical in shortening the desensitization period, allowing patients to tolerate an increased initial dose. Although the reaction rates were not significantly different between the two groups, this was likely due to omalizumab-treated subjects receiving much higher doses of PN protein⁸⁹. If applied to a typical, longer OIT protocol, significant improvements in safety could be demonstrated. Recent studies are consistent with these findings where treatment with omalizumab during OIT conferred a rapid dose escalation and desensitization 90-92. These changes are marked by a reduction in serum levels of allergen-specific IgE antibodies and an increase in allergen-specific IgG4 antibodies ^{90–92}. In one study, decreasing the dosage of omalizumab treatment increased the frequency of OIT-mediated adverse events, leaving only 48% of participants successful in the continuation of the maintenance dose ⁹⁰. In another study, discontinuation of omalizumab during the maintenance phase left only 54% of participants continuing with OIT ⁹¹. The remainder discontinued therapy because of adverse events.

For patients with PN allergy, the evidence tends to favour allergenavoidance in comparison to OIT, as adverse events are quite common. The significant increase in the risk of anaphylaxis and non-anaphylactic reactions may decrease patient adherence to treatment and increase in patient withdrawal. Combination of OIT with an adjunctive treatment, such as omalizumab, holds potential in reducing the frequency of adverse events. Therefore, further trials exploring alternative treatment regimens and allergen delivery systems are required to optimized patient safety.

1.9 SLIT

SLIT involves the daily administration of allergen extract or tablets placed underneath the tongue and held for approximately two-three minutes ^{58,73}. The amount of allergen is then increased to achieve a maintenance dose of approximately 3-7 mg per day, which can be continued indefinitely ^{58,73}. SLIT has demonstrated efficacy in reducing symptom scores (e.g., allergic rhinitis) and medication use for the treatment of seasonal allergic rhinitis due to aeroallergens such as tree, grass, and ragweed pollen ^{93,94}. In the case of birch pollinosis, SLIT reduced both the symptom and medication scores by more than 50% over the 3year course of treatment ⁹⁵. Comparable results were found in patients with allergic rhinitis, with or without intermittent asthma, receiving either SLIT or placebo for 3 years ⁹⁶. Treatment with SLIT resulted in an average of 50% reduction in symptom scores and significantly reduced bronchial hyperreactivity ⁹⁶. In grass pollen allergy, a longitudinal DBPC study randomized subjects to receive either a grass allergen tablet or placebo daily ⁹⁷. Treatment began 16 weeks prior to the start of grass pollen season and continued throughout the season ⁹⁷. After the first season of treatment, patients demonstrated a 30% reduction in rhino-conjunctivitis symptom score, and a 38% reduction in rhino-conjunctivitis medication score in comparison to placebo⁹⁷. Despite 44% of patients having moderate and 56% severe
grass pollen allergy, determined by daily symptom scores during the grass pollen season, the treatment was well tolerated with a withdrawal rate of less than 4% of participants ⁹⁷. Additionally, no serious local side effects or systemic adverse events were reported, as most pertained to mild reactions, including swelling of the tongue or an itchy throat ⁹⁷. After 2 consecutive grass pollen seasons, the reduction in rhino-conjunctivitis symptom and medication scores were maintained in SLITtreated in comparison to placebo ⁹⁸. Furthermore, patients receiving SLIT had a 33% improvement in their quality of life, determined using Juniper's Rhinoconjunctivitis Quality of Life Questionnaire ⁹⁸. Treatment was well tolerated with less than 1% of participants withdrawing due to adverse events ⁹⁸. Allergen-specific IgE increased initially, but continued to decrease over time, consistent with the findings from other allergen immunotherapies ⁹⁸. Interestingly, the levels of allergen-specific IgG4 linearly increased over the course of the treatment, resulting in a 23-fold increase over the span of 2 years ⁹⁸. These findings remain consistent with those of OIT, revealing the importance of IgG4 as a marker of desensitization for allergen immunotherapies.

Although not yet approved for the treatment of food allergy, SLIT is a novel allergen delivery approach, capable of causing desensitization in various clinical trials. One of the first published cases was reported in 2003 for a patient allergic to kiwi fruit ⁹⁹. This patient suffered severe anaphylaxis upon the consumption of kiwi, including 3 episodes of allergic shock accompanied by a loss of consciousness and subsequent hospitalization ⁹⁹. The patient was severely allergic, as even small traces

of kiwi left on a knife used to prepare a desert at a restaurant elicited anaphylaxis ⁹⁹. The patient underwent a modified-SLIT protocol, where he/she received diluted kiwi extract 3 times a day, increasing the amount of allergen with subsequent doses ^{99,100}. The patient increased their tolerance to kiwi from 10⁻⁵ mg to 1 mg of kiwi pulp prior to reaching a 1 cm³ cube of fresh or frozen kiwi maintained for 5 years ^{99,100}. Immunoglobulin analysis demonstrated an increase in the protective IgG4 antibodies, and a decrease in kiwi-specific IgE antibodies after treatment ^{99,100}. Interestingly, despite cessation of therapy for 4 months, the patient was able to tolerate a 1 cm³ cube of fresh kiwi without any adverse reactions upon resuming immunotherapy ¹⁰⁰. This case encouraged the exploration of SLIT for the treatment of food allergy.

One of the first randomized DBPC trials to examine the efficacy and tolerance of SLIT in food allergy was applied to hazelnut allergy ¹⁰¹. Patients were randomly assigned to receive either hazelnut immunotherapy or placebo, and efficacy was determined by a DBPC food challenge after 8-12 weeks of treatment ¹⁰¹. Approximately 95% of patients in the active group (receiving SLIT) were able to reach the maximal dose of 11.56 g from a baseline tolerance of ~2.29 g over the course of 4 days ¹⁰¹. Additionally, approximately 50% of patients receiving active treatment were able to reach the highest dose of the study, 20 g, within the treatment period ¹⁰¹. Despite administering doses in the range of grams, only 0.2% of administered doses resulted in mild reactions, such as itchiness or swelling of the

throat ¹⁰¹. Immunoglobulin and cytokine analysis from blood samples revealed an increase in IgG4 and IL-10 after immunotherapy, only in the active group ¹⁰¹.

Similar results were obtained from the treatment of peach allergy with SLIT. With 6 months of peach SLIT, the active group had a 3-9-fold increase in the amount of peach tolerated, and 5.3 times decrease in SPT with no serious adverse events reported ¹⁰². Immunoglobulin analysis demonstrated an increase in Pru p 3, the major peach allergen, specific IgG4¹⁰². Interestingly, one murine study examined the effect of peach-SLIT in combination with a CpG motif-containing oligodeoxyribonucleotide (ODN-CpG) as an adjuvant to induce a Th1/Treg specific response ^{103,104}. Mice were intranasally sensitized with Pru p 3 and lipopolysaccharide (LPS) ^{103,104}. Mice then underwent 8 weeks of SLIT + ODN-CpG treatment and then challenged at 1- or 3-weeks following treatment cessation ^{103,104}. Treated mice were protected from a drop in core body temperature and demonstrated a decrease in Prup3-specific IgE and IgG1 antibody levels following treatment ^{103,104}. Additionally, an increase in Treg cells and intracellular levels of IL-10/IFN- γ cytokines within CD4⁺ splenocytes was noted ^{103,104}. This protection was maintained for 3 weeks after stopping treatment ^{103,104}. Therefore, combining the SLIT treatment with an adjuvant, such as ODN-CpG, to induce a shift away from the TH2 phenotype may be a beneficial route for future treatments.

In an early non-placebo-controlled study of children with cow's milk allergy, 6 months of milk SLIT increased the threshold of milk tolerated from 39 to 143 mL ¹⁰⁵. In another study, children with cow's milk allergy were subjected to

either SLIT alone (7 mg daily) or SLIT followed by OIT at 2 different doses (1000 mg or 2000 mg daily) ¹⁰⁶. At 12 and 60 weeks of maintenance, the children were challenged with 8 g of milk protein ¹⁰⁶. The study revealed that SLIT followed by OIT was far more effective in comparison to SLIT alone (60-80% versus 10% in passing of the 8 g milk protein challenge) ¹⁰⁶. However, 40% of children across both groups regained reactivity as early as 1 week upon treatment cessation ¹⁰⁶. A decrease in the SPT results and an increase in cow's milk-specific IgG4 levels were observed in all groups ¹⁰⁶. Systemic reactions were more common in the OIT groups with antihistamine usage in up to $\sim 11\%$ of doses in comparison to $\sim 1.5\%$ of SLIT ¹⁰⁶. Although not as efficacious, SLIT maintains the high degree of safety in food allergy treatment, as seen in the treatment of aeroallergens. A limitation of this study pertains to the dosage protocol of SLIT. The OIT groups escalated to doses in the range of grams, whereas the SLIT group was continued on 7 mg for the period of 60 weeks. The challenge was administered at a dose only 4-8 times larger for the OIT groups and at a dose ~1142 times larger for SLIT. The volume of dosage that can be applied under the tongue further poses an obstacle, as exceeding this threshold would most likely result in ingestion of allergen. Therefore, SLIT may benefit from an extended dose escalation, to help decrease the fold difference between the maintenance dose and the OFC dose. Together, these reasons may be why the SLIT group had a lower success rate in comparison to OIT.

SLIT has also been examined for both efficacy and safety in the treatment of PN allergy. In one randomized controlled trial, PN-allergic patients were

27

administered either PN-SLIT or placebo for 44 weeks ¹⁰⁷. A 5 g OFC was conducted after 44 weeks, and subjects successfully consuming 5 g or 10-fold more PN powder from baseline were considered responders ¹⁰⁷. 14/20 (70%) of SLIT-treated patients and 3/20 (15%) of placebo-treated patients successfully passed the oral food challenge ¹⁰⁷. The median tolerated dose increased from 3.5 to 496 mg in the SLIT-treated group ¹⁰⁷. Furthermore, this tolerated dose increased to 996 mg after an additional 24 weeks of SLIT¹⁰⁷. The 10,855 doses were well tolerated with 95.2% of them being symptom-free, and the rest being limited to oropharyngeal itching or tingling ¹⁰⁷. This degree of safety was upheld after 3 years of treatment where 98% of administered doses were well tolerated without any severe reaction or use of epinephrine ¹⁰⁸. However, an OFC at an increased dose of 10 g of PN powder at the 3-year mark yielded only 4 participants who successfully tolerated this dose ¹⁰⁸. Furthermore, these 4 patients exhibited sustained unresponsiveness upon passing a secondary 10 g OFC 8 weeks after treatment cessation ¹⁰⁸. Since the participants were unable to tolerate the high dose, this study would probably have benefited from a secondary dose escalation where the likelihood of participants tolerating a higher dose during the OFC would potentially increase. In another DBPC study subjects underwent a 6-month dose escalation, followed by 6 months of maintenance with PN-SLIT¹⁰⁹. The treated group was able to successfully ingest 20 times more PN protein than the placebo group (median dose: 1710 mg versus 85 mg) ¹⁰⁹. A decrease in SPT and basophil activation after PN-stimulation was observed in the treated group ¹⁰⁹. As seen with other immunotherapies, PN-specific

IgE levels initially increased during the escalation phase prior to decreasing over the remaining months ¹⁰⁹. In contrast, PN-specific IgG4 levels increased over the course of treatment ¹⁰⁹. IL-5 levels were decreased after treatment; however, no statistically significant changes were reported for IL-13, IL-10, Treg cells, or IFN- γ measured from the supernatants of PN-stimulated PBMCs ¹⁰⁹. Continuation of PN-SLIT at 2 mg/day for 3-5 years resulted in 86.5% of participants successfully consuming >750 mg and 62% of participants successfully consuming >1750 mg of PN-protein during the OFC¹¹⁰. Furthermore 32.4% (12/37) of participants passed a 5,000 mg OFC, a dose 2,500 times larger than that of maintenance 110 . Of the 12 who passed the higher-dosed challenge, 10 subjects (27% of treated patients) achieved sustained unresponsiveness after discontinuing treatment for 2-4 weeks ¹¹⁰. Over the course of treatment, the median PN-specific IgE levels decreased from 83.9 kU/L (range, 7.7-1636 kU/L) to 20.0 kU/L (range, 1.6-1051.8 kU/L), and to 28.0 kU/L (range, 10.3-219) for those who achieved sustained unresponsiveness ¹¹⁰. The median PN-specific IgG4 increased from 0.3 mg/L (range, 0-13.1 mg/L) to 10.9 mg/L (range, 0-231.0 mg/L) at the end of the study ¹¹⁰. The IgG4:IgGE ratio increased from ~ 1.45 to ~ 356 , demonstrating induction of protective antibodies ¹¹⁰. Doses were well tolerated with no epinephrine use ¹¹⁰. Side effects were reported for 4.8% of doses, with the most common symptom being oropharyngeal itching ¹¹⁰. Extending PN-SLIT for a long-term protocol does seem to increase the threshold of tolerated allergen. However, these studies could benefit from examining how well patients are able to tolerate additional dose escalation after the

maintenance period, to help increase the threshold of tolerated allergen. In conclusion, published literature demonstrates the capabilities of SLIT in desensitizing patients, while maintaining a high degree of safety.

A limitation to SLIT is that, in comparison to OIT, it is not as efficacious at achieving desensitization. In a double-blinded study, PN-allergic children were randomized to receive either SLIT with placebo OIT, or OIT with placebo SLIT ¹¹¹. Maintenance doses was 3.7 mg/day for SLIT and 2000 mg/day for OIT ¹¹¹. OFCs were conducted at 6 and 12 months of maintenance ¹¹¹. Subjects in both groups demonstrated >10-fold increase in the tolerated dose at 12 months ¹¹¹. OIT achieved a 141-fold increase in the allergen tolerance threshold, while SLIT only achieved a 22-fold increase ¹¹¹. A decrease in IgE levels and an increase in IgG4 were seen in both groups, although to a greater extent in OIT¹¹¹. Therefore, current studies demonstrate that OIT exceeds SLIT in terms of efficacy; however, it is accompanied by more systemic side effects ^{106,111}. Indeed, systemic reactions occurred in 42.8% of doses for PN-OIT (requiring epinephrine, antihistamines, or β -agonists; and treatment withdrawals) and in only 9% of the doses for SLIT ¹¹¹. 3/11 participants in the OIT group discontinued the treatment due to side effects, versus 1 who received SLIT¹¹¹. For the 10 participants who initially received SLIT and were then switched to OIT after 12 months, 2 discontinued the treatment due to side effects ¹¹¹. It is evident that although SLIT may not be as efficacious as OIT, it does deliver the allergen with a high degree of clinical safety and patient adherence. In conclusion, SLIT presents a novel approach for the delivery of food

allergens capable of inducing desensitization, while upholding a high degree of safety. Additional studies using differing dosing protocols and comparisons to other immunotherapies will be beneficial in determining the applicability of SLIT to the general food allergic population.

1.10 Thesis Objective

A number of immunotherapy approaches are being investigated for the treatment of food allergy, although OIT remains the gold standard. Despite demonstrating high efficacy, it is accompanied by severe systemic side effects that discourage patients from continuing treatment ^{58,73,84}. The failure to achieve sustained unresponsiveness in most patients following treatment discontinuation suggests a potential role for combinational therapies to induce lasting protection 58,73 . The end goal of this investigation was to evaluate the impact of SLIT + anti-IL-4R α combination therapy in two murine models of food allergy. However, there were several steps and optimizations required prior to reaching this goal. Specifically, we first established a comprehensive understanding on the capabilities of SLIT in inducing clinical and cellular changes in our murine models of food allergy. In contrast to other models of murine food allergy, the models used here are those of severe food-induced anaphylaxis, and thus, evaluate the capabilities of SLIT under extreme conditions ¹¹². Additionally, our models allow us to examine the effect of SLIT treatment on the generation of IgE. In the epicutaneous model (Figure 1A), only allergen-specific IgG1 antibodies emerge after the tape-stripping protocol, while allergen-specific IgE antibodies are produced only after the nonsensitizing s.c. allergen exposures. In the intragastric model (Figure 1B), IgE production is only measurable after the non-sensitizing intragastric allergen reexposures. For these studies, we utilized chicken egg protein ovalbumin (OVA) as our model antigen, which we have demonstrated produces immune and physiological responses analogous to PN; also, selecting OVA as our model antigen allows us to eventually use novel allergen-specific cell purification methods for future experiments not covered within this thesis. Lastly, we conducted this experimentation in 3 phases: Phase 1) Prophylactic effects of SLIT alone, Phase 2) Semi-therapeutic/Therapeutic effects of SLIT alone, and Phase 3) Investigation of SLIT + anti-IL-4R α as a combinational therapy.

This MSc thesis sought to first optimize SLIT protocols in two models of food allergy and, secondly, to investigate the efficacy of SLIT in modulating clinical and humoral responses in a prophylactic and semi-therapeutic setting.

Chapter 2: Methods

2.1 Animals

Six to eight weeks old female C57BL/6 mice were purchased from Charles River Laboratory (Ottawa, Ontario). Mice were housed in a pathogen-free environment and maintained on a 12-hour light/dark cycle. All experiments outlined were approved by the Animal Research Ethics Board of McMaster University. A total of 295 mice were used for the experiments reported in this document.

2.2 Murine models of allergic sensitization

Epicutaneous Sensitization

Mice were subjected to a previously reported protocol, which is outlined in Figure 1A ^{41,113,114}. Backs of mice were first trimmed with electric clippers, followed by shaving with a mechanical razor to remove any remaining hair (Figure 2A). Water could be used to help ease the shaving process and prevent irritation of the skin. If redness was seen during this process, shaving was stopped and moved onto another area on the back, as the development of scabs would hinder the tape stripping process. Next, using a piece of tape, wrapped around the fingers, the outermost skin layer was gently peeled off (Figure 2B). The back became slightly red, with no hair on the shaven area (Figure 2C). Next, 20 μ L of chicken egg protein, ovalbumin (OVA) (Albumin from chicken egg white, A5378, Millipore Sigma, St Louis, USA) at 10 mg/mL was directly applied onto the shaven back (Figure 2D). This process was repeated Monday-Friday for 10 consecutive days.

Two weeks after the final tape stripping session, mice received 3 s.c. administrations of 100 μ g OVA in 0.5 mLs of PBS, Monday, Wednesday, and Friday. Serum was collected prior to the s.c. re-exposures, and one day prior to the systemic allergic challenge. One day prior the systemic challenge, mice received 500 μ g of anti-CD16/32 (Bio X Cell, BE0307, Clone 2.4G2) intraperitoneally as previously published ¹¹⁵.

Gut Sensitization

Mice were subjected to a modified gastric sensitization protocol, outlined in Figure 1B 21,116 . Held in the supine position, mice received an initial priming intragastric gavage of 1 mg OVA and 5 µg cholera toxin (CT, Biological Laboratories) in a final volume of 0.5 mL PBS. After this initial "priming gavage" mice were rested for 4 weeks. Next, mice received non-sensitizing gavages of 1 mg OVA alone once a week for 3 weeks. Two weeks following the last oral gavage, mice underwent a systemic allergen challenge. Serum was collected and stored prior to the OVA alone gavages and one day prior to the systemic allergen challenge.

2.3 SLIT and Oral Tolerance

Mice were held while maintaining a high scruff where both the ears fall in between the thumb and the index finger (Figure 3A). Upon inspection in the supine position, the mouth was open, with little room available for the head movement (Figure 3B). Next, 5 μ L of OVA (20 μ g, 100 μ g or 1,000 μ g) was administered directly underneath the tongue, behind the lower front teeth using a pipette (Figure 3C). Mice were held in the supine position, parallel to the floor, for 20 seconds, and then released back into the cage (Figure 3D). SLIT was done at the same time every Monday to Friday.

For the oral tolerance comparator group utilized in Phase 1 of experimentation, mice received 10 intragastric gavages of 1 mg OVA twice a week for 5 weeks.

2.4 Systemic Allergen Challenge

The challenge was conducted with 5 mg of OVA in 500 μ L of PBS injected intraperitoneally into sensitized mice. Mice were then observed over the course of 40 minutes for clinical signs of anaphylaxis. Core body temperatures were recorded at 10-minute intervals using a rectal probe digital thermometer (VWR). Clinical symptoms were graded on a scale from 1-5 (1 = in-ear scratching, 2 = reduced motion, 3 = motionless, 4 = no response to whisker stimulus, 5 = prodding or seizure) as outlined in Table 1. Hemoconcentration was recorded by bleeding mice at 40 minutes post systemic challenge using a heparin coated microhematocrit capillary tube and measured using the Hemastat microhematocrit centrifuge (EKF Diagnostics PLC, San Antonio, USA). All mice received 1 mL of warmed PBS intraperitoneally and placed on a heating pad for 4-5 hours. Mice were then examined for recovery post-challenge and either left on the heating pad overnight or taken off and returned to the housing rack.

2.5 Serum Collection

Mice were first anesthetized with isoflurane and then peripheral blood was collected by bleeding from the retroorbital plexus using non-heparinized capillary tubes. For Phase 1 of experimentation, mice were bled: 1) prior to the allergen re-exposures, and 2) one day prior to the systemic allergen challenge (Figure 4A, Figure 7A). Mice in Phase 2 of experimentation were bled: 1) after the tape stripping protocol, or priming gavage, 2) After the cessation of the SLIT protocol, and 3) one day prior to the systemic challenge (Figure 10A, Figure 16A). Peripheral blood was spun at 4°C at 13,000 rpm (397 rcf) for 7 minutes, and serum was pipetted out into a separate Eppendorf tube. Serum was stored at -20°C.

2.6 Splenocyte Collection and Culture

Three days post-allergen challenge, mice were anesthetized, and spleens were harvested. Spleens were crushed through a 40-µm nylon strainers with Hanks buffer into a Petri dish within a biosafety cabinet. Samples were then spun at 1,240 rpm (361 rcf, 4°C, break on) for 10 minutes and shaken with ACK lysis buffer for 90 seconds to lyse red blood cells. ACK reaction was slowed down with 20-30 mL of Hanks and samples were spun at 1,240 rpm (361 rcf), 4°C, break on.

Splenocytes for culture were resuspended with sterile filtered RPMI (cRPMI) containing 10% FBS, 1% L-glutamine, 1% penicillin/streptomycin, and 55 μ mol/L 2-mercaptoethanol (Gibco, Grand Island, USA). Samples underwent cell counts and were diluted to a final concentration of 8,000,000 live cells/mL (in cRPMI) and plated onto flat bottom 96-well plates at 100 μ L/well (800,000

cells/well) in triplicates for both stimulated and unstimulated groups. Wells then received either 100 μ L of cRPMI or 100 μ L of OVA at 0.5 mg/mL. Samples were placed in the incubator for 5 days. On day 5, supernatants were collected from samples (with pooled triplicates) and stored at -80°C until further analysis.

2.7 ELISAs

OVA-specific IgG1

First, 96-well NUNC Maxi-Sorp plates (Thermo Fisher Scientific, Waltham, USA) were coated with 4 μ g/mL OVA in carbonate-bicarbonate buffer (made in-house at 50 nM, Millipore Sigma, St Louis, USA) at 100 µL/well and incubated overnight at 4°C. Next morning, plates were aspirated, and blocked with 1% BSA/PBS at 100 µL/well for 2 h at room temperature. During the incubation, samples were prepared by dilution to 1/20, 1/200, 1/2000, 1/2000. Plates were washed 3 times and incubated with the prepared samples overnight at 4° C. On the third day, the plates were washed 3 times, and then coated with biotinylated goat anti-mouse IgG1 (Southern Biotech, Birmingham, USA) at 50 µL/well and incubated for 2 h at room temperature. After washing 3 times, plates were incubated with alkaline-phosphatase streptavidin (Sigma) for 1 h at room temperature. The assay was developed with *p*-nitrophenyl phosphate tablets (Sigma) in a solution of 4-parts water to 1-part 5x diethanolamine buffer (Sigma) and stopped with 2 M NaOH. Absorbance was read at 405 nm via Multiskan FC photometer (ThermoFisher Scientific).

OVA-specific IgE

First, 96-well NUNC Maxi-Sorp plates (Thermo Fisher Scientific) were coated with 2 µg/mL rat anti-mouse IgE (Clone: R35-72, BD Pharmigen, San Jose, USA) at 50 µL/well. Next, standards were prepared by diluting dig-oxygenated OVA (ANP technologies, Newark, USA, kit 90-1023-1KT) 8 times from 62.5 ng/mL to 0 ng/mL, by a factor of 2 during each dilution. Standards were plated in duplicates at 50 µL/well and the plates were incubated at 4°C overnight. Coated plates were washed 3 times and blocked with 5% skim milk powder in PBS for 1 h in a 37°C warm room. After washing 3 times, samples (diluted at 1:2 and 1:8 in 1% skim milk) were transferred to the plate and stored at 4°C overnight. The next day, plates were washed 5 times and incubated with 50 µL/well dig-oxygenated OVA (300 ng/mL, diluted in 1% skim milk) (excluding the standard columns) for 90 minutes at room temperature. Plates were washed 5 times and anti-digoxigenin-POD (poly) Fab fragments (Roche, Basel, Switzerland, 5 μ L in 0.3% BSA/PBS) was added to the plate at 50 μ L/well and incubated for 1 h at room temperature, covered from light. Plate was washed 5 times and the colour reaction was developed using TMB liquid substrate (3,3',5,5'-Tetramethylbenzidine, Sigma) and stopped with 2M H2SO4. Absorbance was read at 450 nm.

2.8 Luminex

Frozen supernatants from splenocyte cultures were tested in the MILLIPLEX Immunology Multiplex Assays (MCYTOMAG-70K kit, Millipore Sigma, St Louis, USA) and analyzed through the MAGPIX XMAP Technology system (Luminex, Austin, USA). Samples were evaluated for the secretion of IFN- γ , IL-3, IL-4, IL-5, IL-9, IL-10, IL-12 (p70), IL-13, and IL-17. Cytokines for statistical analysis were included if they fell within the detectable range (3.2-10000 pg/mL).

2.9 Statistics

All data were analyzed using GraphPad Prism (v.6.0) and displayed as mean \pm SEM. A one or two-way analysis of variance (ANOVA) with a Turkey's or Bonferroni's *post hoc* test were used for data analysis. Differences were considered significant when the p-value was less than 0.05 (*, °).

Chapter 3: Results



Figure 1. Schematics of models of allergic sensitization. **A**, epicutaneous sensitization with IgG1 generation occurring after tape stripping, and IgE being generated upon subcutaneous re-exposures. **B**, intragastric sensitization with mice producing IgE only after the OVA-alone intragastric exposures. S.c., subcutaneous; CT, cholera toxin; OVA, ovalbumin.



Figure 2. Visual representation of epicutaneous sensitization adapted from Jiménez-Saiz *et al* 41 . **A**, representation of back after shaving. **B**, tape stripping





Figure 3. Visual representation of the SLIT administration protocol. A, maintain a high scruff when picking up the mouse, so that the ears are encompassed between the thumb and index fingers. **B**, ensure that the mouth is open and the tongue steady. **C**, administer 5 μ L of OVA underneath the tongue, behind the front teeth. **D**, hold the mouse in the supine position for 20 seconds prior to release.



Figure 4. Prophylactic SLIT protects mice against epicutaneous allergic sensitization and allergen-induced clinical reactivity. **A**, schematic for the prophylactic SLIT prior to epicutaneous sensitization. **B**, core body temperature during systemic allergen challenge. **C**, hemoconcentration at 40 minutes post challenge. **D**, clinical scores during systemic challenge (scoring described in Table 1). Data are representative of 1 experiment with 5-10 mice per group, plotted as mean \pm SEM in comparison to the buffer SLIT group (*) or naïve group (°). *, ° P<0.05. Buffer, PBS SLIT; oral, oral tolerance model, s.c., subcutaneous, OVA, ovalbumin.

Evaluation of prophylactic OVA-SLIT in establishing protection against epicutaneous sensitization

Our first aim was to examine the impact of OVA-SLIT in a prophylactic setting. This served two purposes: 1) to examine whether the sublingual route can be used to induce tolerance, thus protecting the mice from any subsequent allergic sensitization, and 2) to validate the SLIT technique and dosage protocol to be used in subsequent experiments. C57BL/6 mice were first administered three weeks of OVA-SLIT, Monday to Friday, at either 1,000 µg (high dose) or 100 µg (low dose)

(Figure 4A). Next, mice received 10 days of tape stripping, Monday to Friday, as shown in Figure 2. Serum was collected two weeks after the final tape strip, followed by 3 non-sensitizing, subcutaneous OVA exposures administered Monday, Wednesday, and Friday. Additionally, a group of mice underwent 10 intragastric gavages, twice a week for 5 weeks, to serve as a model of oral tolerance. This allowed us to compare the effects of sublingual exposure to a more conventional exposure to induce tolerance. Mice were rested for two weeks, prior to the final serum collection. At that point, mice were treated with an anti-CD16/32 monoclonal antibody to prevent the induction of IgG1-mediated anaphylaxis. One day after the serum collection, mice were challenged intraperitoneally with 5 mg of OVA and monitored for clinical signs of anaphylaxis (core temperature, hemoconcentration and clinical scores).

As expected, the PBS buffer-treated allergic mice had a significant drop in core body temperature over the course of 40 minutes, indicative of an IgE-mediated systemic shock (Figure 4B). One mouse in the buffer group was a non-responder, an atypical event in this model (Appendix 1A). The 100 µg SLIT-treated group exhibited a variable, yet intermediate level of protection (Figure 4B). Notably, these mice showed some drop in core body temperature up to the 20-minute mark, at which it plateaued at the 30-minute mark and begun to recover at the 40-minute mark (Figure 4B, Appendix 1A). Examination of individual mice showed that 6/10 mice had returned to baseline temperature at the 40-minute mark, 2/10 had an intermediate drop in temperature, and 2/10 were still fully responsive (Appendix

1A). In contrast, the 1 mg SLIT-treated group exhibited full protection from a drop in core body temperature at all time points and was not significantly different than the naïve group (Figure 4B). One mouse from this group had an intermediate drop in core body temperature for the first 30 minutes but proceeded to recover thereafter (Appendix 1B). Interestingly, the oral tolerance group exhibited an intermediate level of protection, similar to that of the 100 μ g SLIT-treated group (Figure 4B). Individual temperatures showed 2/5 mice exhibiting full protection and 3/5 mice an intermediate level of protection (Appendix 1C). The naïve mice exhibited no drop in core body temperature (Figure 4B).

When undergoing anaphylaxis, mice also exhibit an increase in hemoconcentration due to vascular leakage. Buffer-treated mice exhibited the largest increase in hemoconcentration in comparison to naïve mice (Figure 4C). Consistent with the drop in core body temperature, the 100 µg SLIT-treated group exhibited a significantly less increase in hemoconcentration in comparison to the positive control (Figure 4C). The variability seen in the data reciprocate the temperature graphs, with the highest increase in hemoconcentration corresponding to the mice exhibiting the largest drop in core body temperature. Next, the 1 mg SLIT-treated group exhibited hemoconcentration, similar to that of the naïve group (Figure 4C). The orally exposed mice exhibited a variable but statistically significant lesser increase in hemoconcentration to buffer-treated mice (Figure 4C). A third measure of reactivity was observation of clinical responses (Table 1). We utilize a 5-point based grading scheme, ranging from 0: no clinical signs, to 5: prodding or seizure. The buffer-treated mice exhibited clinical scores upwards of 4, indicating no whisker responses (Figure 4D). Next, the 100 µg SLIT-treated group demonstrated significantly better clinical scores in comparison to the positive control (Figure 4D). Within this group, 2/10 mice exhibited reduced motion, 4/10 mice exhibited in-ear scratching and 4/10 mice exhibited no symptoms, thus demonstrating a variable, intermediate level of protection (Figure 4D). Mice in the 1 mg SLIT-treated group were fully protected except for one mouse that exhibited in-ear scratching; 9/10 mice exhibited no symptoms (Figure 4D). This group was not statistically different than the naïve group. The orally exposed mice showed a variable response with 1/5 mice demonstrating reduced motion, 2/5 with in-ear scratching and 2/5 exhibiting no symptoms (Figure 4D). The naïve group experienced no clinical signs upon allergen challenge (Figure 4D).

In conclusion, administering three weeks of SLIT prophylactically was sufficient to protect mice from severe anaphylaxis upon challenge. Furthermore, the 1 mg dosage provided near full protection in comparison to the 100 μ g dosage of OVA-SLIT.



Figure 5. Prophylactic SLIT reduces the emergence of OVA-specific IgE after epicutaneous sensitization. A-B, serum OVA-specific IgE at two dilutions measured prior to subcutaneous allergen re-exposures, (A) and one day prior to challenge (B). C-D, serum OVA-specific IgG1 at four dilutions measured prior to subcutaneous allergen re-exposures (C) and one day prior to challenge (D). E-F,

comparison of OVA-specific IgE and IgG1 titres across the allergen subcutaneous re-exposures. Data are representative of 1 experiment with 5-10 mice per group, plotted as mean \pm SEM in comparison to the buffer SLIT group (*) or naïve group (°). *, ° P<0.05. Buffer, PBS SLIT; Oral, oral tolerance model; Re-exp, allergen re-exposures. Buffer, PBS SLIT; oral, oral tolerance model; OD, optical density.

Serum was collected from mice at two timepoints (Figure 4A) and analysed for the levels of OVA-specific IgE and IgG1 using a sandwich-ELISA protocol. Post tape stripping samples demonstrated low levels of OVA-IgE amongst all groups, increasing upon subsequent allergen re-exposures (Figure 5A-B). As expected, the buffer-treated group showed the largest increase in OVA-IgE levels following allergen re-exposures (Figure 5E). The 100 µg SLIT-treated group showed a significant reduction in the level of OVA-IgE production post allergen re-exposures in comparison to the positive control (Figure 5E). Next, the 1 mg SLIT-treated group showed only a slight increase in titres of OVA-IgE antibodies at the secondary time point (Figure 5E), but these levels were not significantly different from those of naïve mice. The orally exposed mice demonstrated an increase in OVA-IgE antibodies, not significant from that of the positive control (Figure 5E). However, it is important to note the high degree of variability observed in this analysis, consistent with the variability seen within the hemoconcentration and clinical signs (Figure 4C, D). Additionally, it is plausible that these findings are the result of the protocol itself, 1mg intragastric OVA gavage twice a week for five weeks. Optimizing the frequency of the dose intervals may ameliorate the OVA-IgE production seen within this group. Together, these data suggest that this protocol for oral tolerance was not sufficient in establishing complete protection against epicutaneous sensitization.

Analysis of OVA-IgG1 demonstrated a similar level of induction amongst all groups following tape stripping (Figure 5C-D). At the 1:20 dilution, the 1 mg SLIT-treated mice exhibited a statistically significantly lower level of OVA-IgG1 in comparison to the positive control. These results are in accordance with the epicutaneous model of sensitization, where only IgG1 antibodies are observed prior to subcutaneous allergen exposures ^{41,113,114}. Upon allergen subcutaneous exposures, the buffer-treated, 100 µg SLIT-treated, and 1 mg SLIT-treated mice all showed a similar increase in the levels of OVA-specific IgG1 antibodies (Figure 5F). Interestingly, the oral tolerance model had a significant decrease in levels of OVA-IgG1 following the subcutaneous allergen re-exposures (Figure 5F). It is important to note that the levels of OVA-IgG1 for this group was highly variable. resulting in no statistical difference from the levels of the naïve group (Figure 5F). We believe that the small number of mice allocated to this group may explain the large degree of variability seen within the OVA-IgG1 analysis. With more mice allocated to the group, we expect them to produce levels of OVA-IgG1 antibodies similar to that of the buffer group. Together, these data indicate that prophylactic treatment with SLIT in this model reduces the production of OVA-specific IgE.

Three days following the allergen challenge, we harvested the spleens of mice to investigate the impact of SLIT on the T cell compartment. Harvested splenocytes were cultured *in vitro* with OVA for five days. Supernatants collected

48

on the fifth day were examined for various cytokines through the Luminex cytokine detection assay. Unfortunately, the high degree of variability observed prevents definite conclusions from this analysis. Nevertheless, frozen replicates of the samples have been stored, and will be repeated to achieve a less variable result.

We first investigated the impact of prophylactic SLIT on the TH2associated cytokines, namely, IL-4, IL-5, and IL-13. An induction of IL-4 with OVA was seen within both SLIT-treated groups, the oral tolerance group, and the positive control (Appendix 3A). A similar trend was demonstrated for IL-5 and IL-13 upon stimulation with OVA (Appendix 3B-C). Therefore, this suggests an induction of a TH2 response in the SLIT-treated mice, despite the observed significant protection from clinical reactivity upon challenge. Next, we sought to examine any changes in the regulatory cytokine, IL-10 and the Th1 cytokine, IFN- γ . No differences were observed for the induction of IFN- γ (Appendix 3D) although the naïve group does seem to have some induction of IFN- γ (Appendix 3D). However, IL-10 was significantly upregulated in both the SLIT-treated groups with OVA stimulation (Appendix 3E). IL-3, IL-9, IL-12 (p70), and IL-17 were also examined during this analysis; however, these cytokines remained unchanged upon OVA stimulation within all groups (Appendix 3F-I).



Figure 6. Prophylactic SLIT protects mice against intragastric sensitization and food-induced clinical reactivity. **A**, schematic for prophylactic SLIT prior to intragastric sensitization. **B**, core body temperature during challenge. **C**, hemoconcentration at 40 minutes post challenge. **D**, clinical scores during challenge (as described in Table 1). Data are representative of 1 experiment with 5-10 mice per group, plotted as mean \pm SEM in comparison to the buffer SLIT group (*) or naïve group (°). *, ° P<0.05. OVA, ovalbumin; CT, cholera toxin; Buffer, PBS SLIT; oral, oral tolerance model.

Evaluation of prophylactic OVA-SLIT in establishing protection against intragastric sensitization

Next, we evaluated the impact of prophylactic OVA-SLIT on subsequent intragastric allergic sensitization. C57BL/6 mice were first administered three weeks of OVA-SLIT, Monday to Friday, at either 1,000 (high dose) or 100 μ g (low dose) (Figure 6A). Next, the mice received a single priming gavage of OVA + CT, a known TH2 adjuvant ¹¹⁷. The mice were rested for four weeks, followed by serum collection prior to allergen exposures. A non-sensitizing, OVA alone intragastric

gavage was administered once a week for three weeks. Serum was collected two weeks after the final gavage, followed by an allergen challenge with 5 mg of OVA the next day.

Like the epicutaneous model, the buffer-treated mice exhibited a significant drop in core body temperature upon challenge (Figure 6B). One non-responder (1/10 mice) in the buffer group resulted in large error bars (Figure 6B, Appendix 2A-C). The low dose 100 µg SLIT-treated group exhibited a partial level of protection (Figure 6B). These mice had an intermediate drop in core body temperature, plateauing at 20-30 minutes, and then showing a trend towards recovery (Figure 6B). Within this group, 2/10 mice were fully reactive, 2/10 mice were partially protected while demonstrating a trend towards recovery after 30 minutes, and 6/10 mice had no clinical response (Appendix 2A). The high dose 1 mg SLIT-treated group also responded similarly to the epicutaneous model (Figure 6B). Here, 9/10 mice were fully protected from clinical reactivity, and one mouse had a partial response, but recovered back to baseline after 20 minutes (Figure 6B, Appendix 2B). Interestingly, the orally exposed group was, in this model, fully protected from clinical reactivity, with all mice exhibiting no drop in core body temperature, similar to the naïve group (Figure 6B, Appendix 2C).

Analysis of hemoconcentration data demonstrated a statistically significant reduction in the SLIT-treated, and orally exposed groups (Figure 6C). The 100 µg SLIT-treated group showed variability, as seen in the drop in core body temperature

(Figure 6C). The 1 mg SLIT-treated group, and the orally exposed group showed low variability and results not different than those of naïve mice (Figure 6C).

With respect to clinical signs, we observed more severe reactions in the positive control group in comparison to the epicutaneous model (Figure 6D). Within the buffer group, 3/10 mice achieved the highest clinical score indicating seizure or prodding, 4/10 mice had no response to whisker stimulation, and 2/10 mice had reduced motion (Figure 6D). The scores were significantly reduced in the 100 µg SLIT-treated mice, with 1/10 mice undergoing seizure, 1/10 mice having reduced motion, 2/10 mice with in-ear scratching and 6/10 exhibiting no clinical symptoms (Figure 6D). The 1 mg SLIT-treated group had a less variable response with only 3/10 mice reacting with in-ear scratching, and 7/10 mice with no clinical response (Figure 6D). The orally exposed group was similar with 2/5 mice exhibiting in-ear scratching and 3/5 undergoing no clinical symptoms (Figure 6D).



Figure 7. Prophylactic SLIT reduces the emergence of OVA-specific IgE antibodies after intragastric sensitization. **A-B**, serum OVA-specific IgE antibodies at two dilutions measured prior to subcutaneous allergen re-exposures (**A**) and one day prior to challenge (**B**). **C-D**, serum OVA-specific IgG1 antibodies at four dilutions measured prior to intragastric allergen re-exposures, (**C**) and one day prior

to challenge (**D**). **E-F**, comparison of OVA-specific IgE and IgG1 antibody titres across the allergen intragastric exposures. Data are representative of 1 experiment with 5-10 mice per group, plotted as mean \pm SEM in comparison to the buffer SLIT group (*) or naïve group (°). *, ° P<0.05. OD, optical density; OVA, ovalbumin; CT, cholera toxin; Buffer, PBS SLIT; Oral, oral tolerance model; Re-exp, allergen re-exposures.

As expected, analysis of OVA-specific IgE levels in serum revealed no IgE production following the initial intragastric priming gavage (Figure 7A). IgE production was observed only at the second serum collection, *i.e.*, after the non-sensitizing OVA alone intragastric exposures (Figure 7B). The positive control group demonstrated a substantial increase in OVA-IgE production following allergen re-exposures (Figure 7E). The 100 µg SLIT-treated group demonstrated an intermediate, yet significant production of OVA-IgE in comparison to the positive control (Figure 7E). Consistent with no drop in core body temperature, both the 1 mg SLIT-treated group and the orally-exposed group had no increase in IgE production across both serum collection timepoints (Figure 7E). The levels of IgE in these groups were not statistically different from naïve mice (Figure 7E).

Previous data from our lab showed that following a single priming gavage, mice do not generate any measurable antibody response, as demonstrated by the buffer group and 100 µg SLIT-treated group (Figure 7C). Interestingly, both the 1 mg SLIT-treated, and oral tolerance group show a small, but significant level of OVA-IgG1 production following the priming gavage (Figure 7C). Upon the allergen re-exposures, the buffer- treated group had a large increase in OVA-IgG1 production (Figure 7D, F). The 100 µg SLIT-treated group had an increase in OVA- IgG1 levels, although statistically lower than the levels of the buffer control group (Figure 7D, F). In the 1 mg SLIT-treated and oral tolerance group, there was little to no change in the levels of OVA-IgG1 (Figure 7D, F). The lack of change in the production of OVA-IgG1 was unique and may be a feature of establishing tolerance in the gut sensitization model.

Three days after the systemic allergen challenge, we harvested spleens to set up splenocyte cultures culture as mentioned previously. Since both the epicutaneous and priming model were analyzed at the same time, the following data also contain a high degree of variability, preventing us from drawing any definite conclusions. However, the trends observed in this set of data may provide some insight on markers of interest for future experimentation. Nonetheless, replicates have been stored and the analysis will be repeated to achieve a less variable response.

First, with respect to TH2 associated cytokines, we observed an induction of IL-4 by both the buffer and 100 μ g SLIT-treated group upon OVA stimulation (Appendix 4A). However, upon stimulation with OVA, both the 1 mg SLIT-treated, and oral tolerance group showed no change in IL-4 secretion (Appendix 4A). As IL-4 is required for IgE class switching, the lack of it explains why both groups had no IgE production prior to challenge, and subsequent clinical reactivity. IL-5 seems to be induced in both the buffer-treated and 100 μ g SLIT-treated groups (Appendix 4B). The 1 mg SLIT-treated, and oral tolerance group lacked an IL-5 response (Appendix 4B). Induction of IL-13 was seen to an extent in the buffer group, but all other groups showed no response (Appendix 4C). In sum, we see a trend suggesting an inhibition of TH2 cytokines in the groups that exhibited low clinical reactivity upon challenge. No firm conclusions can be drawn with respect to IFN- γ and IL-10 due to the high variability. No changes were observed with OVA stimulation in IL-3, IL-12 (p70), and IL-9 (Appendix 4F-H). IL-17 was significantly induced in the buffer group and to a small degree in the oral tolerance group, but not in any other group (Appendix 4I).



Figure 8. Semi-therapeutic SLIT (6/9 wk) partially protects epicutaneous sensitized mice from food-induced clinical reactivity. **A**, schematic for semi-therapeutic SLIT administration following tape-stripping. **B**, core body temperature during allergen challenge. **C**, hemoconcentration at 40 minutes post challenge. **D**, clinical scores during challenge (as described in Table 1). Data are representative of 1 experiment with 5-10 mice per group, plotted as mean \pm SEM in comparison to the buffer SLIT group (*) or naïve group (°). *, ° P<0.05. SLIT no-re, SLIT treated mice that received no allergen re-exposures. OVA, ovalbumin; s.c., subcutaneous; Buffer, PBS SLIT; NO-RE, no allergen re-exposures.

Evaluation of semi-therapeutic OVA-SLIT in establishing protection against epicutaneous sensitization

The second phase of experimentation evaluated the efficacy of SLIT as a semi-therapeutic treatment. Here, C57BL/6 mice underwent the 10-day tape stripping protocol prior to receiving OVA-SLIT treatment (Figure 8A). Noted earlier, this model offers a unique feature in that the production of OVA-specific IgE only occurs after the 3 non-sensitizing subcutaneous OVA exposures. Therefore, we can examine whether administering OVA-SLIT prevented the transition from IgG1 to IgE. We denote the treatment as being semi-therapeutic, since the mice are sensitized as indicated by the presence of OVA-specific IgG1 and the capacity to generate an IgE response upon subsequent non-sensitizing exposures. We chose 1 mg of OVA to be the optimal dosage for SLIT, as it demonstrated the greatest level of protection in the prophylactic model. Additionally, we hypothesized that 3 weeks could be too short and, thus, extended it arbitrarily to 6 and 9 weeks. The 6-week SLIT-treated group was adjusted to start 3 weeks after the 9-week group. We also added a group that received 9 weeks of OVA-SLIT, but no subcutaneous OVA re-exposures (SLIT no-re), to examine if SLIT itself could induce OVA-IgE. Mice were rested for 2 weeks following SLIT, prior to receiving the OVA subcutaneous exposures (Figure 8A). An allergen challenge was conducted 2 weeks following the last subcutaneous exposure (Figure 8A).

The buffer group exhibited a significant drop in core body temperature, while the naïve mice showed no clinical responses (Figure 8B). Administering 6 weeks of OVA-SLIT resulted in partial protection (Figure 8B). Within this group, 5/10 mice had a partial drop in core body temperature prior to an increase towards recovery at the 30-minute mark (Appendix 5A). The response was variable with some mice recovering, others maintaining a partial level of protection and the rest reacting no differently than that of buffer control (Appendix 5A). Extending OVA-SLIT to 9 weeks statistically improved the level of protection (Figure 8B). Despite the degree of variability, the mean core body temperature of the 9-week OVA-SLIT group was ~36°C, in comparison to ~33°C for 6-weeks of OVA-SLIT and ~30°C for buffer control (Figure 8B, Appendix 5B). Closer analysis showed 4/10 mice with no change at the 40-minute mark in core body temperature in comparison to baseline, 5/10 mice with a partial response, and 1 mouse that initially had a 4° C drop in core body temperature but recovered back to baseline after 20-minutes (Appendix 5B). The SLIT no re-exposure group showed a small drop in core body temperature which recovered back to baseline at 40-minutes (Figure 8B). Although the temperature measurements were not statistically different than those of the naïve group (Figure 8B), there was some variability (Appendix 5C). This indicates that SLIT by itself can induce a slight degree of reactivity.

Examination of hemoconcentration data demonstrated a partial increase in comparison to the buffer group when treated with either 6 or 9 weeks of OVA-SLIT (Figure 8C). Despite variability, 9 weeks of OVA-SLIT had most mice clustered

towards a hemoconcentration similar to that of the naïve group (Figure 8C). Both the 9-week OVA-SLIT treated and SLIT no re-exposure group had a similar level of hemoconcentration to that of naïve mice (Figure 8C).

Clinical scores followed a similar trend as seen with the other outcomes. The buffer group had the highest score, with 3/10 mice having seizures/prodding, 2/10 mice motionless, and 5/10 mice exhibiting reduced motion (Figure 8D). Clinical scores were significantly better in all SLIT-treated groups (Figure 8D). In the 6-week SLIT-treated group, 9/10 mice exhibited only reduced motion and 1/10 mice had no response (Figure 8D). For the 9-week SLIT-treated group, 6/10 mice had reduced motion, 2/10 mice displayed in-ear scratching, and 2/10 mice had no clinical response (Figure 8D). Within the SLIT no-re-exposure group, 4/10 mice experienced in-ear scratching, and no clinical signs were observed in the naïve group (Figure 8D). Collectively, these findings indicate that 9 weeks of OVA-SLIT provides a superior protection compared to 6 weeks.


Figure 9. Semi-therapeutic SLIT partially prevents the emergence of OVA-specific IgE antibodies after epicutaneous sensitization. A-C, serum OVA-specific IgE antibodies at two dilutions measured after the tape stripping (A), after the cessation of SLIT (B), and one day prior to challenge (C). D-F, serum OVA-specific IgG1 antibodies at four dilutions measured after tape stripping (D), after the cessation of SLIT (E), and one day prior to challenge (F). G-H, comparison of OVA-specific IgE and IgG1 antibody titres across the measured timepoints. Data are representative of 1 experiment with 5-10 mice per group, plotted as mean \pm SEM in comparison to the buffer SLIT group (*) or naïve group (°). *, ° P<0.05. OD, optical density; Buffer, PBS SLIT; NO-RE, no allergen re-exposures, Re-exp, allergen re-exposures.

Serum was collected at 3 time points: 1) after the 10 days of tape stripping, 2) after the 6-9 weeks of OVA-SLIT, and 3) one day prior to allergen challenge (Figure 8A). Following tape stripping, a small amount of OVA-IgE production is only seen in the 6-week SLIT-treated group, although at extremely low levels (Figure 9A). Post SLIT treatment, we saw low levels of OVA-IgE antibodies in all SLIT-treated groups (Figure 9B). Upon subcutaneous exposures, the buffer, 6- and 9-week SLIT-treated groups all showed IgE production (Figure 9C). Low levels of OVA-IgE are noted for the SLIT no re-exposures group, consistent with the slight reactivity seen during the allergen challenge (Figure 8D). Examining the change in IgE levels over the three serum collection points, the 6-week SLIT group produced levels of OVA-IgE no different than the buffer group (Figure 9G). The 9-week SLIT group demonstrated an intermediate level of OVA-IgE, significantly lower than both the buffer and 6-week SLIT groups (Figure 9G). The SLIT no reexposures treated group maintained low levels of OVA-IgE across the serum collections (Figure 9G).

As expected, all tape-stripped mice (excluding naïve mice), had an OVA-IgG1 response (Figure 9D). Treatment with OVA-SLIT resulted in an increase in the levels of OVA-IgG1 in comparison to the buffer group (Figure 9E, H). Upon the administration of subcutaneous OVA re-exposures, all groups, excluding naïve mice, produced OVA-IgG1 (Figure 9E, H). Within the SLIT-treated groups, administration of OVA subcutaneous re-exposures did not increase the levels of OVA-IgG1, but rather the levels were sustained from the post-SLIT serum collection (Figure 9H). The data indicate that the levels of OVA-specific IgG1 for both the 6- and 9-weeks of SLIT treatment were not significantly different from that of the buffer control.



Figure 10. Impact of semi-therapeutic SLIT (12/15 wk) on food-induced clinical reactivity in epicutaneously sensitized mice. **A**, schematic for semi-therapeutic SLIT administration following tape-stripping. **B**, core body temperature during systemic allergen challenge. **C**, hemoconcentration at 40 minutes post systemic challenge. **D**, clinical scores during systemic challenge (as described in Table 1). Data are representative of 1 experiment with 5-10 mice per group, plotted as mean \pm SEM in comparison to the buffer SLIT group (*) or naïve group (°). *, ° P<0.05. OVA, ovalbumin; s.c., subcutaneous; Buffer, PBS SLIT; NO-RE, no allergen re-exposures.

Evaluation of extended semi-therapeutic OVA-SLIT in establishing protection following epicutaneous sensitization

Following the previous set of experiments, we sought to further improve the level of protection offered by semi-therapeutic OVA-SLIT. Our results established that 9 weeks of OVA-SLIT provided a higher degree of protection in comparison to 6 weeks. Therefore, we hypothesized that extending the duration of SLIT to 12-15 weeks would provide better protection. This set of experimentation followed the same procedures as mentioned previously, except for a longer duration of OVA SLIT (Figure 10A).

All mice in the buffer group had a drastic drop in core body temperature reaching an average of ~29.5°C (8.5-degree change) at 40-minutes in comparison to the negative control (Figure 10B). Surprisingly, both SLIT-treated groups exhibited a significant drop in core body temperature, although slightly but significantly less than the buffer group (Figure 10B). The drop in core body temperatures between the two SLIT-treated groups was not significant (Figure 10B). Unlike previous data, we did not observe recovery in the SLIT-treated groups, and most mice became clinically reactive (Appendix 6A, B). The SLIT no re-exposures group had one mouse that fully responded, and one mouse that had a partial drop in core body temperature (Appendix 6C). The variability amongst the SLIT no re-exposures group was noticeably less in comparison to the previous set of experimentation. Buffer and the 12-week SLIT group had similar levels of hemoconcentration (Figure 10C). The 15-week SLIT group had a significantly lower hemoconcentration, although a large degree of variability was seen (Figure 10C). The SLIT no re-exposures group demonstrated variable responses but, as a group, the response was not significantly different than that of the naïve group (Figure 10C).

The buffer, 12-week SLIT, and 15-week SLIT all had statistically similar clinical scores (Figure 10D). Within the buffer group, 2/10 mice had seizures/prodding, 3/10 had no whisker response, 1/10 was motionless, 2/10 were reduced, and 1 had in-ear scratching (Figure 10D). Within the 12-week SLIT group, 1/10 mice underwent seizure/prodding in, 2/10 had no whisker response, 1/10 was motionless, and 6/10 had reduced motion (Figure 10D). Within the 15-week SLIT group, only 1/10 experienced seizure/prodding, and 9/10 had reduced motion (Figure 10D). One mouse in the SLIT no re-exposures group had a seizure, but the remainder had no clinical response (Figure 10D). When examining the clinical scores, 15-weeks of OVA-SLIT appeared to better protect the mice against clinical reactivity, demonstrated by the large proportion of mice with lower clinical scores in comparison to 12-weeks of SLIT (Figure 10D). We expected OVA-SLIT to improve the level of protection following an extended duration, but this was not the case. Clearly, the drop in core body temperature and clinical scores indicate that the 12- and 15-weeks of OVA-SLIT was not protective.



Figure 11. Impact of semi-therapeutic SLIT on the emergence of OVA-specific IgE and IgG1 after epicutaneous sensitization. A-C, serum OVA-specific IgE antibodies at two dilutions measured after the tape stripping (A), after the cessation of SLIT (B), and one day prior to challenge (C). D-F, serum OVA-specific IgG1 antibodies at four dilutions measured after tape stripping (D), after the cessation of SLIT (E), and one day prior to challenge (F). G-H, comparison of OVA-specific IgE and IgG1 antibody titres across the measured timepoints. Data are representative of 1 experiment with 5-10 mice per group, plotted as mean \pm SEM in comparison to the buffer SLIT group (*) or naïve group (°). *, ° P<0.05. OD, optical density; Buffer, PBS SLIT; NO-RE, no allergen re-exposures, Re-exp, allergen re-exposures.

Analysis of the immunoglobulins demonstrated no increase in OVA-IgE in all groups at the post-tape stripping serum collection point, as expected (Figure 11A). Following the administration of OVA-SLIT, IgE only significantly increased in the 12-week OVA-SLIT group (Figure 11B). Interestingly, we did not see any induction of OVA-IgE in the 15-week OVA-SLIT-treated group (Figure 11B). The levels of OVA-IgE at this timepoint were extremely low, so lack of significance could be due to variability between groups. Buffer, 12-week, and 15-week SLITtreated groups all produced OVA-IgE upon subcutaneous allergen re-exposures (Figure 11C). The levels of OVA-IgE were not significantly different amongst the three groups (Figure 11C, G). Over the course of the three bleeding points, the 12week SLIT group had a significantly higher amount of OVA-IgE in comparison to the positive control (Figure 11G). Although not significant in comparison to the naïve group, we detected low levels of OVA-IgE in the SLIT no re-exposures group, possibly explaining the variable number of intermediate responses observed during the allergen challenge (Figure 11G).

All groups, except naïve mice, generated OVA-IgG1 upon the completion of tape stripping (Figure 11D). After the administration of OVA-SLIT, all SLITtreated groups showed an increase in OVA-IgG1 (Figure 11E). The largest change in OVA-IgG1 production across these two bleeding points was seen in the 12-week OVA-SLIT-treated group (Figure 11E, H). Following the allergen re-exposures, OVA-IgG1 production was detected amongst all groups, although to a significantly lower extent in the SLIT no re-exposures group (Figure 11F). The SLIT no reexposures treated group maintained a similar level of OVA-IgG1 across all three bleeding points (Figure 11H). Overall, semi-therapeutic OVA-SLIT induced an OVA-IgG1 response, although no differences were observed in the production of OVA-IgG1 at the final bleed point (Figure 11H).



Figure 12. Experimental repeat of the 9-week semi-therapeutic SLIT protocol confirms protection from food-induced clinical reactivity in epicutaneously sensitized mice. A, schematic for semi-therapeutic SLIT administration following tape-stripping. B, core body temperature during systemic allergen challenge. C, hemoconcentration at 40 minutes post challenge. D, clinical scores during challenge (as described in Table 1). Data are representative of 1 experiment with 5-10 mice per group, plotted as mean \pm SEM in comparison to the buffer SLIT group (*) or naïve group (°). *, ° P<0.05. OVA, ovalbumin; s.c., subcutaneous; Buffer, PBS SLIT; NO-RE, no allergen re-exposures.

Confirmation of the protection following epicutaneous sensitization and subsequent clinical reactivity by 9-weeks of semi-therapeutic OVA-SLIT

It was intriguing to find that 12 and 15 weeks of OVA SLIT conferred less protection than 9-weeks. Therefore, we conducted a repeat of the 9-week OVA SLIT group to confirm our initial findings (Figure 12A). This experiment followed the exact protocol as mentioned in Figure 10A, with the exclusion of the 6-week OVA-SLIT group.

Upon challenge, we saw very similar results in the drop in core body temperature, hemoconcentration, clinical scores, change in OVA-IgE and IgG1 in comparison to the initial experiment (Figure 8-9, 12-13, Appendix 7). Pooled results consistently demonstrated that 6 and 9 weeks of SLIT treatment provided partial protection against food-mediated clinical reactivity (Appendix 10).



Figure 13. Impact of 9-week SLIT treatment on the emergence of OVA-specific IgE in epicutaneously sensitized mice. A-C, serum OVA-specific IgE at two dilutions measured after the tape stripping (A), after the cessation of SLIT (B), and one day prior to challenge (C). D-F, serum OVA-specific IgG1 antibodies at four dilutions measured after tape stripping (A), after the cessation of SLIT (B), and one day prior to challenge (C). G-H, comparison of OVA-specific IgE and IgG1 antibody titres across the measured timepoints. Data are representative of 1 experiment with 5-10 mice per group, plotted as mean \pm SEM in comparison to the buffer SLIT group (*) or naïve group (°). *, ° P<0.05. OD, optical density; Buffer, PBS SLIT; NO-RE, no allergen re-exposures, Re-exp, allergen re-exposures.

Analysis of the immunoglobulin response was consistent with the initial set of experimentation. As expected, little to no OVA-IgE is detected in all groups following tape stripping (Figure 13A). Interestingly, a small amount of OVA-IgE was only detected in the SLIT no re-exposures group following the conclusion of SLIT treatment (Figure 13B). Data pooled with the initial experiment were consistent, with all SLIT-treated groups exhibiting production of OVA-IgE following SLIT treatment, and one day prior to challenge (Appendix 10D). However, the OVA-IgE one day prior to challenge in the 9 week SLIT-treated group was significantly lower than those in the buffer and the 6 week SLIT-treated group (Appendix 10D). Serum levels of OVA-IgG1 remained consistent with the initial findings amongst all groups (Figures 9H, 13H, Appendix 10E). Overall, the results of this experiment confirm our initial findings indicating that 9 weeks of semi-therapeutic OVA SLIT was partially protective against food-mediated clinical reactivity.



Figure 14. Semi-therapeutic SLIT treatment for 9 weeks following intragastric sensitization was not protective against food-induced clinical reactivity. **A**, schematic for semi-therapeutic SLIT administered following the initial priming intragastric gavage. **B**, core body temperature during allergen challenge. **C**, hemoconcentration at 40 minutes post challenge. **D**, clinical scores during challenge (as described in Table 1). Data are representative of 1 experiment with 5-10 mice per group, plotted as mean \pm SEM in comparison to the buffer SLIT group (*) or naïve group (°). *, ° P<0.05. OVA, ovalbumin; CT, cholera toxin; Buffer, PBS SLIT; NO-RE, no allergen re-exposures.

Evaluation of semi-therapeutic OVA-SLIT in establishing protection

following intragastric sensitization

The second part of phase 2 entails evaluating the efficacy of semitherapeutic SLIT in a gut model of allergic sensitization. Here, C57BL/6 mice first received a single intragastric priming gavage of OVA + CT (Figure 14A). Mice were rested for four weeks prior to the administration of OVA-SLIT. As mentioned earlier, the mice do not generate a detectable OVA-specific antibody response following the priming oral gavage; however, IgE and IgG1 are both detected upon

the subsequent OVA-alone intragastric exposures. Like the epicutaneous model, the priming model allows us to examine whether OVA-SLIT is capable in preventing the generation of IgE in primed mice. The prophylactic gut model indicated 1 mg of OVA as the optimal protective dose for SLIT treatment, and thus, all SLIT-treated mice received this dose. Additionally, the duration of SLIT was increased to 9-weeks, as we hypothesized that a longer treatment could be required for establishing protection. Two weeks after the final administration of SLIT, mice received a non-sensitizing OVA-alone intragastric gavage, once a week, for a total of three weeks (Figure 14A). An allergen challenge was conducted two weeks following the final OVA gavage. During our initial experimentation, we had two groups, 6 and 9 weeks, for the duration of SLIT; however, this experiment failed due to variable responses in our positive control. Therefore, we used a second set of mice that had completed the priming gavage and were initially scheduled as a repeat. Due to the number of mice available at that time, we decided it was best to only include the 9-week SLIT group.

We found that treating mice with 9 weeks of OVA-SLIT was not protective in this setting (Figure 14B). The buffer group had a significant drop in core body temperature, with 1 mouse being a non-responder (Figure 14B, Appendix 8A). Within the SLIT-treated group, 8/9 mice responded to a similar degree as the positive control, and only 1 mouse had no drop in core body temperature (Figure 14B, Appendix 8A). The SLIT no re-exposures group had 8/10 mice respond to the same degree as the positive control and only 2/10 mice showed no drop in core body temperature (Figure 14B, Appendix 8B). Consistent with the drop in core body temperatures, we observed that all groups, excluding naïve mice, had a similar increase in hemoconcentration and, thus, no different from each other (Figure 14C). Overall, 9 weeks of 1 mg OVA-SLIT did not prevent anaphylactic reactions.

Despite all groups demonstrating clinical scores not statistically significant from each other, we did observe variance in the SLIT-treated mice (Figure 14D). Within the buffer group, 7/10 mice had seizures/prodding, 1/10 had no whisker response, 1/10 had reduced motion, and 1/10 was a non-responder (Figure 14D). In the OVA-SLIT group, 2/9 mice had seizures/prodding, 4/9 mice had no whisker response, 2/9 were reduced and 1/9 had in-ear scratching (Figure 14D). In comparison to the positive control, a lower proportion of mice reached the highest clinical score, and thus, possibly indicating some protection conferred from SLIT. Within the SLIT no re-exposures group, 7/10 mice had seizures/prodding, 1/10 had reduced motion and 2/10 had no clinical symptoms. Together, these data indicate that 1 mg of SLIT following intragastric sensitization was not protective, suggesting that further optimization of dosage and duration was required.

73



Figure 15. Semi-therapeutic SLIT does not prevent the emergence of OVA-specific IgE and IgG1 after intragastric sensitization. **A-C**, serum OVA-specific IgE at two dilutions measured after the priming gavage (**A**), after the cessation of SLIT (**B**), and one day prior to systemic challenge (**C**). **D-F**, serum OVA-specific IgG1 at four dilutions measured after the priming gavage (**D**), after the cessation of SLIT (**E**), and one day prior to challenge (**F**). **G-H**, comparison of OVA-specific IgE and IgG1 levels across the measured timepoints. Data are representative of 1 experiment with 5-10 mice per group, plotted as mean \pm SEM in comparison to the buffer SLIT group (*) or naïve group (°). *, ° P<0.05. OD, optical density; Buffer, PBS SLIT; NO-RE, no allergen re-exposures; Re-exp, allergen re-exposures.

Serum was collected at three different timepoints: 1) four weeks after the initial priming gavage, 2) after the conclusion of SLIT treatment, and 3) one day prior to the systemic allergen challenge. As expected, no OVA-IgE was detected amongst all groups following the priming gavage (Figure 15A). OVA-IgE levels within the SLIT-treated groups were detected prior to the OVA-alone gavages (Figure 15B). The amount of OVA-IgE detected was not significantly different between the two SLIT groups. Upon the non-sensitizing OVA gavages, the buffer, OVA-SLIT and SLIT no re-exposures groups all had detectable levels of OVA-IgE (Figure 15C). Across the serum collection timepoints, treatment with SLIT resulted in the production of OVA-IgE, at amounts similar to the buffer group prior to challenge (Figure 15G). Similar levels of OVA-IgE between the SLIT-treated and buffer groups is consistent with the clinical findings.

No OVA-IgG1 was detected following the priming gavage within all groups (Figure 15D). Both SLIT-treated groups had a significant increase in OVA-IgG1 production following SLIT treatment, in comparison to the buffer group (Figure 15E, H). Following the OVA gavages, all groups, excluding naïve mice, showed detectable levels of OVA-IgG1 (Figure 15F, H). Interestingly, both SLIT-treated groups showed significantly more OVA-IgG1 antibodies in comparison to the buffer group, unlike what was observed in the semi-therapeutic SLIT, skin sensitized mice (Figure 15F, H). It is uncertain whether this change is meaningful or is the result of variance seen between these models. We do not suspect that the



difference in OVA-specific IgG1 levels had any impact on the clinical results of the systemic allergen challenge.

Figure 16. Lower dosed semi-therapeutic SLIT partially protects intragastric sensitized mice form food-induced clinical reactivity. **A**, schematic for semi-therapeutic SLIT administration following priming gavage. **B**, core body temperature during allergen challenge. **C**, hemoconcentration at 40 minutes post challenge. **D**, clinical scores during challenge (as described in Table 1). Data are representative of 1 experiment with 5-10 mice per group, plotted as mean \pm SEM in comparison to the buffer SLIT group (*) or naïve group (°). *, ° P<0.05. OVA, ovalbumin; CT, cholera toxin; NO-RE, no allergen re-exposures.

Lowering the dose of OVA-SLIT confers partial protection against foodinduced clinical reactivity within intragastric sensitized mice

The previous set of data demonstrated that treatment of primed mice with 9 weeks of 1 mg OVA-SLIT was not protective against food-induced clinical reactivity. Therefore, we sought to examine treatment with lower doses of SLIT, specifically 100 and 20 µg of OVA. Additionally, we hypothesized that increasing

the duration of SLIT with a lowered dosage could be required to achieve protection. With that in mind, we assessed the 100 μ g dose for 9 and 12 weeks, and the 20 μ g dose for 9 weeks. Additionally, we included two separate groups for the SLIT no re-exposures group, one for the 100 μ g dose, and one for the 20 μ g dose, both 12 weeks. Mice followed a similar protocol as the previous experiment, outlined above (Figure 16A).

The buffer control group had a drastic drop in core body temperature, with 9/10 mice dropping to a mean temperature of ~30.4°C (Figure 16B). One mouse was a non-responder and showed no drop in core body temperature (Figure 16B, Appendix 9A). Treatment with the 9-week 100 µg OVA-SLIT protocol was partially protective as the mean temperature drop was 34.6°C at 40 minutes (Figure 16B). Here, 3/10 mice had a severe response, 3/10 mice had an intermediate drop in core body temperature, and 4/10 mice had no drop in core body temperature (Appendix 9A). Extending this SLIT treatment to 12-weeks provided similar protection (Figure 16B). Here, 3/10 mice had a severe response no different than buffer, 2/10 mice had a partial response, recovering after 20 minutes, 1/10 had an initial drop in temperature, but proceeded to recover back to baseline at 40 minutes, and 4/10 mice had no drop in core body temperature (Appendix 9B). Decreasing the dose of SLIT to 20 µg for 12 weeks was still protective, demonstrating an intermediate drop in core body temperature, reaching ~33.9°C at 40 minutes (Figure 16B). Within the 20 μ g SLIT group, 2/10 mice had a severe drop in core body temperature, 4/10 had an intermediate drop, and 4/10 had no drop in core body temperature (Appendix 9C). The 12-week 20 μ g SLIT group performed similarly to the 9-week 100 μ g SLIT treated group. The 100 μ g SLIT no re-exposures group did have a statistically significant drop in core body temperature lower than the buffer group, reaching ~35.1°C at 40 minutes (Figure 16B). Here, 2/5 mice had a complete response, 1/5 had a partial response, recovering towards baseline after 20 minutes, and 2/5 had no drop in core body temperature (Appendix 9A). The 20 μ g SLIT no re-exposures group demonstrated a similar level of protection from a drop in core body temperature, although this was not statistically significant in comparison to the buffer group due to a large standard error (Figure 16B). Here, 2/5 mice had a complete response, 1/5 had an intermediate drop, and 2/5 had no drop in core body temperature (Appendix 9E).

Measurement of hemoconcentration yielded large variability amongst the groups. The buffer group had the highest level of hemoconcentration, which was not statistically significant from among all groups, excluding naïve and the 12-week 100 µg SLIT-treated group (Figure 16C). All SLIT-treated groups had a high degree of variability in their hemoconcentration, consistent with the variability seen within the drop in core body temperatures. The 12-week 100 µg SLIT group had many mice clustering towards the lower end of hemoconcentration, suggesting an optimal dose protocol (Figure 16C).

Clinical signs were consistent with findings in core temperature and hemoconcentration. Within the buffer group, 4/10 mice had seizures/prodding, 2/10 had no whisker response, 3/10 were motionless and 1 was a non-responder (Figure

16D). The 12-week 100 µg SLIT-treated mice were the only group with a significantly lower clinical score in comparison to the buffer control (Figure 16D). Within the 12-week 100 µg SLIT-treated group, 1/10 mice had a seizure/prodding, 2/10 mice had reduced motion, 4/10 mice had in-ear scratching, and 3/10 mice had no clinical symptoms (Figure 16D). In comparison to the buffer, we observed an increase in the number of mice clustering in the 1-2 clinical score range, suggesting protection by the SLIT treatment. The 9-week 100 µg SLIT-treated group had 1/10 mice experience seizure/prodding, 1/10 had no whisker response, 4/10 had reduced motion, 1/10 had in-ear scratching and 3/10 had no symptoms (Figure 16D). Similarly, the 12-week 20 µg SLIT-treated group had 2/10 mice experience seizure/prodding, 1/10 had no whisker response, 2/10 had reduced motion, 1/10 had in-ear scratching, and 4/10 had no symptoms (Figure 16D). For the 100 µg SLIT no re-exposures had 1/5 mice with no whisker response, 1/5 had reduced motion, 1/10with in-ear scratching, and 2/5 with no symptoms (Figure 16D). For the 20 µg SLIT no re-exposures group, 2/5 mice had seizures/prodding, 1/5 had reduced motion, 1/5 with in-ear scratching and 1 had no clinical symptoms (Figure 16D). Therefore, SLIT by itself does induce a slight degree of clinical reactivity. Clearly, the best protection, although partial, was achieved with the 12-week 100 µg SLIT group protocol.



Figure 17. Semi-therapeutic SLIT after intragastric sensitization partially prevents the emergence of OVA-specific IgE. **A-C**, serum OVA-specific IgE at two dilutions measured after the intragastric priming gavage (**A**), after the cessation of SLIT (**B**), and one day prior to challenge (**C**). **D-F**, serum OVA-specific IgG1 at four dilutions measured after the intragastric priming gavage (**D**), after the cessation of SLIT (**E**), and one day prior to challenge (**F**). **G-H**, comparison of OVA-specific IgE and IgG1 levels across the measured timepoints. Data are representative of 1 experiment with 5-10 mice per group, plotted as mean \pm SEM in comparison to the buffer SLIT group (*) or naïve group (°). *, ° P<0.05. OD, optical density; Buffer, PBS SLIT; NO-RE, no allergen re-exposures; Re-exp, allergen re-exposures.

Antibody analysis of serum collected at three different timepoints demonstrated a decrease in the levels of OVA-IgE within the SLIT-treated mice

(Figure 17A-C). Four weeks following the initial intragastric priming gavage, a low level of OVA-IgE was detected in the buffer-treated mice (Figure 17A). This level remained consistent at the second serum collection time point, where all groups exhibited similar levels of OVA-IgE (Figure 17B, G). Prior to the allergen challenge, all groups had detectable levels of OVA-IgE; however, all SLIT-treated groups had a statistically significant lower level of OVA-IgE (Figure 17C). Over the course of treatment, the initial generation of OVA-IgE within the SLIT-treated groups was sustained upon the OVA-alone intragastric allergen exposures (Figure 17G). The decrease in OVA-IgE is consistent with the protection seen during the allergen challenge.

OVA-IgG1 was detected in the buffer group following the initial priming gavage (Figure 17D). All other groups had no detectable OVA-IgG1 at this timepoint (Figure 17D). Upon the administration of SLIT, all SLIT-treated groups showed an increase in the production of OVA-IgG1 (Figure 17E, H). The levels of OVA-IgG1 produced by all SLIT-treated groups, excluding the 20 µg SLIT no reexposures group, were significantly higher than the buffer group (Figure 17E). Following allergen re-exposures, all groups, excluding naïve mice, produced similar amounts of OVA-IgG1 (Figure 17F). SLIT-treated groups maintained the levels of OVA-IgG1 produced across the second and third serum collection points, as seen consistently in our previous SLIT experiments.

Chapter 4: Discussion

A plethora of studies have investigated the efficacy and safety of OIT for the treatment of food allergies. Clearly, OIT can successfully desensitize most patients. However, it is associated with a significant number of adverse events, greater in fact compared to allergen avoidance. Therefore, the decision to engage in OIT requires an informed joint evaluation of risks and benefits by patients, or their guardians, and health care professionals. SLIT is an alternative route of immunotherapy that has demonstrated similar efficacy to OIT but a significantly better safety profile. Most of the available evidence regarding SLIT has been generated in the context of aeroallergies. The primary goal of this MSc thesis was to first optimize and then determine the impact of SLIT in two murine models of food-induced anaphylaxis, a model of epicutaneous sensitization and a model of intragastric sensitization. In addition, SLIT was administered in two distinct settings, prophylactic and semi therapeutic. The terminology "semi therapeutic" was chosen to indicate that SLIT was administered to either primed mice (intragastric model) or mice that had generated an allergen-specific IgG1 response (epicutaneous model) but prior to the development in both instances of an allergenspecific IgE response. The allergen used in these studies was OVA. The reason for choosing this allergen is that future studies, beyond the scope of this thesis, will investigate in depth the impact of SLIT on allergen-specific B cells. These are very rare cells and to circumvent this limitation, our laboratory has developed a tetramer technology to harvest enriched populations of these cells. At this point, only fully validated tetramers for OVA are available. It is important to note that, in these models, OVA yields similar outcomes as PN.

Prophylactic OVA-SLIT prevents epicutaneous and intragastric allergic sensitization, and subsequent food-mediated anaphylaxis.

In Phase 1 of the research plan, we optimized the sublingual delivery of OVA as an approach to potentially establishing tolerance in two murine models of food allergy and anaphylaxis (Figure 1A-B). We found that prophylactic treatment with 3 weeks of OVA SLIT at 1 mg completely prevented the generation of OVAspecific IgE in both models. As clinical and physiological outcomes in these models are largely dependent on IgE, SLIT-treated mice were, expectedly, fully protected from anaphylaxis (Figures 4 and 7). A limitation of the allergen-specific immunoglobulin analysis is that a serum collection timepoint immediately after the conclusion of SLIT was not included. This timepoint would have allowed a direct examination of the immunoglobulin response elicited by SLIT versus allergic sensitization. For comparison, we included a group of mice that were subjected to a conventional oral tolerance protocol ^{118–120}. Interestingly, we observed that these mice were completely protected from anaphylaxis in the intragastric sensitization model, but only partially in the epicutaneous sensitization model. The OVAspecific IgE data was concordant with the clinical readouts. We hypothesize that this divergence could be due to the distinct routes of treatment and sensitization. The draining sites in epicutaneous and intragastric sensitization are the inguinal and the mesenteric lymph nodes respectively. Thus, it could be argued that a preexisting tolerant response, induced by the oral gavages, in the mesenteric lymph nodes was sufficient to fully prevent the development of allergic sensitization in the same lymph nodes. In contrast, it could have been not sufficient to fully prevent a competing allergic response established in the inguinal lymph nodes, ultimately resulting in partial protection.

A limitation of these experiments is that we did not have an opportunity to examine the length of protection conferred by prophylactic SLIT in these models of food allergy. In other words, whether the mice would remain protected against allergic sensitization at 3- or 6-months following prophylactic SLIT treatment remains unknown. However, these experiments are currently being pursued by other members of the lab that are continuing this project.

To examine the mechanisms underlying SLIT-induced tolerance, we conducted a cytokine analysis of supernatants from OVA-stimulated splenocytes from SLIT-treated mice. Unfortunately, the high variability of the data prevents any firm conclusions. Within the epicutaneous model we observed an apparent induction of a TH2 response in SLIT-treated mice (Appendix 3A-C). These data were in sharp contrast with the clinical results, where the 1 mg SLIT-treated mice had essentially no reactivity upon allergen challenge (Figure 4). However, we also observed a trend towards an induction of IL-10, a regulatory cytokine (Appendix 3E). In contrast, we observed an absence of a TH2 response in the SLIT-treated, intragastric sensitized mice, and an induction of IL-10 (Appendix 4A-C, E). Together, these data may suggest that the SLIT-treated mice in the epicutaneous

model are undergoing two different responses, while the mice in the intragastric model lean towards one type of response. At this point, the mechanisms underlying clinical tolerance in SLIT-treated mice remain to be elucidated. Upcoming work from Lundsgaard *et al.* has examined the capacity and effects of SLIT with Ara h 6, a major PN allergen, in preventing allergic sensitization ¹²¹. Similar to the protocol described here, mice received 3 weeks of SLIT prior to intraperitoneal sensitization ¹²¹. Not only were the mice protected from Ara h 6 sensitization in a dose-dependent manner, but tissue analysis revealed an expansion of Treg cells within the sublingual lymph nodes and blood ¹²¹. In combination with the trends in IL-10 induction observed in our prophylactic settings of SLIT, we hypothesize that Treg cells and IL-10 may play a critical role in the induction of tolerance. The source and specific mechanism behind this production of IL-10 warrants further investigation in future phases of this project.

Semi-therapeutic OVA-SLIT provides protection against food-mediated anaphylaxis in both the epicutaneous and intragastric models of allergic sensitization

In the second phase of experimentation, we sought to examine the semitherapeutic effects of SLIT in the two models of food allergy and anaphylaxis. In the epicutaneous model, SLIT was administered immediately after the tape stripping *i.e.*, at the time mice had already generated an OVA-specific IgG1 response, but before the allergen re-exposures. Therefore, this experimental design was asking whether SLIT would prevent the transition from an IgG1 to an IgE response. In the intragastric model, SLIT was administered 4 weeks following the priming gavage, but before the allergen re-exposures. Thus, the experiment was asking whether SLIT would prevent the activation of allergic memory.

Semi-therapeutic OVA-SLIT in the epicutaneous sensitization model

We found that 9 weeks of treatment with 1 mg OVA-SLIT provided significant, better than 50%, but partial protection, as indicated by the changes in core body temperature, hemoconcentration and clinical score in comparison to the buffer group. The change in the serum levels of OVA-specific IgE was concordant with the clinical outcome. As the impact of SLIT for 6 weeks was comparatively inferior, we argued that extending SLIT beyond 9 weeks could yield an additional benefit. However, treatment for 12-15 weeks did not improve protection and, in fact, resulted in worsening of all the outcome measurements. The protocol of SLIT administration that we used may explain this unexpected outcome. Typically, human trials of food immunotherapies begin with an extremely low dose, at the microgram level, followed by a dose escalation over a period of weeks to months ⁵⁸. The use of dose escalation protocols is intended to minimize adverse events in the early stages of treatment. In our studies, we used a single high dose of allergen from the beginning and throughout the entire treatment. It is, then, possible that maintaining that high dose of allergen beyond 9 weeks reinforced the pathogenic TH2 response.

Induction of allergen-specific IgG has been suggested as a possible mechanism through which allergen immunotherapy mediates tolerance, notably

86

IgG4 in humans ⁵⁸. In mice, it has been shown that IgG antibodies may induce tolerance to foods through their effects on the inhibitory FcγRIIb receptor on mast cells and basophils ^{122,123}. It is possible that the ratios of OVA-IgE to OVA-IgG1 antibodies varied depending on the duration of SLIT. Unfortunately, ELISAs for both the 6-9 week SLIT treatment and the 12-15 week SLIT treatment were ran at different times and, therefore, we are unable at this time to make a direct comparison of OVA-specific IgE:IgG1 ratios amongst the various groups. This hypothesis can be evaluated by repeating the OVA-specific IgE and IgG1 ELISAs with the samples stored from both experiments. Additionally, it is plausible that other antibody isotypes, such as IgG2a, IgG2b, and IgG2c, which we did not measure at this point, could have contributed to the partial protection observed. In fact, recent and ongoing work by a PhD candidate in the lab has shown an upregulation of OVA-specific IgG2b and IgG2c in this setting.

Semi-therapeutic OVA-SLIT in the intragastric sensitization model

We found that treatment of mice with 9 weeks of SLIT at a dose of 1 mg OVA was not protective against food-mediated anaphylaxis and, expectedly, serum levels of OVA-specific IgE were similar to the buffer group (positive control) amongst all SLIT-treated mice (Figure 15). Importantly, the SLIT-treated mice that received no allergen re-exposures also underwent anaphylaxis and produced OVA-IgE. These findings were in sharp contrast to those in the epicutaneous model, where this group did not show any significant response. These findings suggest the possibility that the SLIT-treated mice were swallowing a portion of the sublingually

administered OVA, which is not totally unexpected. As mentioned previously, the route of sensitization and treatment are anatomically separated in the skin model; however, both routes may be shared in the model of gastrointestinal sensitization, assuming that a portion of the allergen is swallowed during SLIT-treatment. We now know that the administration of a single gavage of OVA plus CT (priming) establishes long-lived allergic memory. Consequently, it is plausible that OVA swallowed during SLIT treatment activated a memory response. With this in mind, we hypothesized that lowering the dose of OVA would minimize the amount of OVA reaching the gastrointestinal tract. Indeed, we found that lowering the dose of OVA-SLIT from 1 mg to, particularly, 100 µg for 12 weeks achieved significant but partial protection from anaphylaxis and a drastic decrease in the levels of serum OVA-specific IgE. SLIT alone, *i.e.*, with no allergen re-exposures, induced only a very slight increase in OVA-specific IgE. These data strengthen the hypothesis that small amounts of OVA ingested during SLIT activated allergic memory in the gut. Although we were unable to achieve complete protection through a single-dose treatment approach, these data emphasize the importance of optimizing both the dosage and duration of treatment.

Summary of data findings and future directions

Currently, food allergy remains a disease without a cure. Despite immunotherapies, such as OIT, demonstrating high efficacy in achieving clinical desensitization, OIT is accompanied by a high rate of adverse effects, and protection that is lost upon discontinuation of treatment in most patients ⁵⁸. The goal

of this thesis project was to optimize and evaluate the use of SLIT in two distinct murine models of food allergy and anaphylaxis. In the first phase of experimentation, we optimized a prophylactic SLIT treatment regime that prevented both epicutaneous and intragastric sensitization. In the second phase, we evaluated OVA-SLIT in semi-therapeutic models. Here, we found that the dosage of OVA and the duration of treatment were unique to each model and were key in optimizing this therapy. However, we were unable to achieve complete protection in these set of experiments. This finding is not entirely surprising, as lack of complete protection from reactivity is commonly seen in food allergen immunotherapies in humans ^{58,101,102,107,108}. Typically, patients who received SLITtreatment demonstrate an increase in the maximum tolerated dose of allergen in comparison to controls; however, these doses are often accompanied by non-severe allergic reactions, such as an irritated throat. Therefore, the partial level of protection seen within our SLIT-models may be characteristic of allergen immunotherapy. Nonetheless, further evaluation of dosage and treatment duration may confer greater protection.

Our data show that SLIT, given as a single constant dose, provides a partial level of protection in a semi-therapeutic setting. It would be valuable to optimize a dose-escalation SLIT protocol in these murine models that mimics human protocols of OIT. Mice could be started on a small dose that is escalated every week until a maximum dose of 1 mg is reached for maintenance. A starting dose of 100 μ g would be appropriate as this dose demonstrated minimal reactivity within our semi-

therapeutic SLIT models of food allergy. The rate of dose escalation would require a degree of optimizations, as increasing the dosage too quickly may cause more harm than benefit. Upon reaching the 1 mg maintenance dose, the mice could continue it for at least 6 weeks, as demonstrated by our semi-therapeutic SLIT data. If successful, the semi-therapeutic SLIT-treated mice would exhibit full protection from clinical reactivity upon the systemic allergen challenge. With an established model, we may then proceed to explore the fundamental mechanisms through which this tolerance is established.

An additional consideration is the incorporation of serum collection at multiple time points, which would allow for the analysis of immunoglobulin kinetics. These mice may be bled every 2 weeks throughout the treatment protocol. Additionally, examining OVA-specific IgG subtypes (IgG2a, IgG2b), IgA, and IgM would provide additional information on how SLIT is impacting humoral immunity. We must also consider the analysis of the B and T cell compartments within the spleens, cervical lymph nodes, inguinal lymph nodes (in the skin model) and mesenteric lymph nodes (in the gut model). If SLIT is inducing lasting tolerance in these models, we hypothesize that there would be a shift away from the allergic TH2 phenotype towards a regulatory phenotype. This can be addressed through flow cytometric analysis of the various tissues following the systemic allergen challenge. Key cells would include TH2 cells, Treg cells, and a complete analysis of all BC isotypes. It may also be of interest to examine a relatively new and unexplored subset of BCs known as IL-10⁺ B regulatory (Breg) cells that play

a role in the suppression of pro-inflammatory responses ¹²⁴. Although relatively unexplored, these cells appear to play crucial roles in the suppression of clinical symptoms via IL-10 and induction of Foxp3⁺ Tregs ^{124,125}. In addition to flow cytometry, Luminex analysis of cytokines from OVA/media stimulated splenocyte cultures of SLIT-treated, allergic, and naïve mice can further assist in the elucidation of how SLIT establishes its protective effects. We would expect to see a decrease in the production of TH2 cytokines (IL-4, -5, -13) and an induction of the regulatory cytokine, IL-10. Together, these set of experimentations could cohesively establish a mechanism by which SLIT exhibits its protective effects and the impact it has on both the T/B cell compartments and humoral immunity.

Lastly, it may be important to examine the role of SLIT in a combinational therapy for the treatment of food allergy within these models. It is evident from current literature that food allergen immunotherapies do confer a high degree of protection but is still accompanied by allergic reactions ⁵⁸. Additionally, the established protection is transient and is often lost upon treatment cessation. Therefore, an alternative form of treatment is required to achieve long-term tolerance, even after treatment cessation.

One strategy would be to combine a biologic, capable of inducing cellular changes within the allergic memory cell compartment, with an allergen immunotherapy to help shift the TH2 response towards a regulatory response. This concept has been previously explored in the context of both peach and cow's milk allergy where treatment with SLIT in combination with an adjuvant (*e.g.*, ODN-

91

CpG, IL-2/anti-IL-2Ab complex antibodies, or Omp16) conferred better, lasting protection in comparison to SLIT alone ^{103,126,127}. Within our models of food allergy, anti–IL-4R α is the ideal candidate for the biologic component of treatment. As demonstrated previously by our lab, treatment with anti–IL-4R α not only abolished IgE generation and clinical reactivity, but this protection was maintained for up to 6 weeks following treatment discontinuation ⁵⁶. We would then hope that treatment with anti–IL-4R α in combination with allergen stimulation could drive the TH2 allergic response towards one of a regulatory phenotype. For administration of allergen, the sublingual route or oral route could both be effective for this combinational therapy. However, it is important to note that within both human and murine studies of food immunotherapy, SLIT, in comparison to OIT, exhibits a higher degree of safety ⁵⁸. In turn, this would likely promote a higher treatment adherence amongst patients, and thus, provide an effective treatment with a superior safety profile.

The first part of my MSc thesis optimized the SLIT protocol within the epicutaneous and intragastric models of food allergy. Here we demonstrated that prophylactic treatment with 3 weeks of 1 mg OVA SLIT was protective against allergic sensitization and subsequent clinical reactivity. The second component focused on the evaluation of SLIT as a semi-therapeutic treatment. Here, we demonstrated that SLIT, when administered for after tape stripping (9 weeks at 1 mg OVA) or the priming gavage (12 weeks at 100 μ g), was only capable of achieving partial protection from food-induced clinical reactivity. The work

presented here provides both an optimized SLIT protocol and evidence for the efficacy of SLIT in the treatment of food allergy. Future work should aim to explore alternative SLIT dosing protocols to achieve complete protection against clinical reactivity when used both as a semi-therapeutic and therapeutic treatment.

Tables

Table 1. Clinical grading scheme for evaluating mice undergoing a systemic allergic challenge

Score	Symptom
1	In ear scratching with hind foot
2	Reduced Motion
3	Motionless
4	No whisker response
5	Prodding and/or seizure





Appendix 1. Group-sorted drop in core body temperature graphs for Phase 1 OVA-SLIT treated, skin sensitized mice.



Appendix 2. Group-sorted drop in core body temperature for Phase 1, OVA-SLIT treated, gut sensitized mice.


Appendix 3. Cytokine analysis of cultured splenocytes from Phase 1, skinsensitized mice. The first bar of each group indicates media-stimulated splenocytes, and the second bar represents stimulation with OVA. Due to sample constraints in the Luminex assay, only one sample from the oral tolerance group was analyzed. **A-C**, analysis of TH2 associated cytokines. **D-E**, analysis of regulatory cytokines IFN- γ and IL-10. **F-I**, analysis of IL-3, IL-9, IL-12 (p70), and IL-17. Data are representative of 1 experiment with 5-10 mice per group, plotted as mean ± SEM relative to the media control per group. *P<0.05.



Appendix 4. Late-phase cytokine analysis of cultured splenocytes from Phase 1, intragastric-sensitized mice. The first bar of each group indicates media-stimulated splenocytes, and the second bar represents stimulation with OVA. **A-C**, analysis of TH2 associated cytokines. **D-E**, analysis of regulatory cytokines IFN- γ and IL-10. **F-I**, analysis of IL-3, IL-9, IL-12 (p70), and IL-17. Statistical significances are shown relative to the media control per group. Data are representative of 1 experiment with 5-10 mice per group, plotted as mean ± SEM relative to the media control per group. *P<0.05.



Appendix 5. Group-sorted drop in core body temperature for Phase 2, 6-9 week, OVA-SLIT treated, skin sensitized mice.



Appendix 6. Group-sorted drop in core body temperature for Phase 2, 12-15 week, OVA-SLIT treated, skin sensitized mice.



Appendix 7. Group-sorted drop in core body temperatures for Phase 2, 9-week OVA-SLIT treated, skin sensitized mice.



Appendix 8. Group-sorted drop in core body temperature for Phase 2, 9-week OVA-SLIT treated, gut sensitized mice.



Appendix 9. Group-sorted drop in core body temperature for Phase 2, OVA-SLIT treated, gut sensitized mice.



Appendix 10. Pooled results of semi-therapeutic SLIT (6/9 wk) demonstrating partial protection against food-induced clinical reactivity in epictuaneous sensitized mice. **A**, Drop in core body temperature during systemic allergen challenge. **B**, Hemoconcentration at 40 minutes post challenge. **C**, Clinical score of reactivity during challenge (as described in Table 1). **D**, Change in serum OVA-specific IgE levels measured after tape stripping, after cessation of SLIT, and one day prior to systemic allergen challenge. **E**, Change in serum OVA-specific IgG1 levels measured after tape stripping, after cessation of SLIT, and one day prior to systemic allergen challenge. Data are represented of 2 identical experiments (n = 20 mice for Buffer, 9wk SLIT, and SLIT no-re groups; n = 10 mice for 6wk SLIT, and naive groups). Statistical significances are shown relative to the buffer SLIT group (*) or naïve group (°). *, ° P<0.05. Buffer, PBS SLIT; NO-RE, no allergen re-exposures; OD, optical density; Re-exp, allergen re-exposures.

References

- 1. Sicherer SH, Sampson HA. Food allergy: A review and update on epidemiology, pathogenesis, diagnosis, prevention, and management. *Journal of Allergy and Clinical Immunology*. Published online 2018. doi:10.1016/j.jaci.2017.11.003
- 2. Wood RA. Food allergen immunotherapy: Current status and prospects for the future. *Journal of Allergy and Clinical Immunology*. Published online 2016. doi:10.1016/j.jaci.2016.01.001
- 3. Osborne NJ, Koplin JJ, Martin PE, et al. Prevalence of challenge-proven IgE-mediated food allergy using population-based sampling and predetermined challenge criteria in infants. *Journal of Allergy and Clinical Immunology*. Published online 2011. doi:10.1016/j.jaci.2011.01.039
- 4. Loh W, Tang MLK. The epidemiology of food allergy in the global context. *Int J Environ Res Public Health*. 2018;15(9). doi:10.3390/IJERPH15092043
- 5. Soller L, Ben-Shoshan M, Harrington DW, et al. Overall prevalence of self-reported food allergy in Canada. *Journal of Allergy and Clinical Immunology*. Published online 2012. doi:10.1016/j.jaci.2012.06.029
- Clarke AE, Elliott SJ, Pierre YS, Soller L, La Vieille S, Ben-Shoshan M. Temporal Trends in Prevalence of Food Allergy in Canada SPAACE View project Coping with Anxiety and Food Allergy Calmly View project. J Allergy Clin Immunol Pract. Published online 2019. doi:10.1016/j.jaip.2019.10.021
- 7. Wu LC, Zarrin AA. The production and regulation of IgE by the immune system. *Nat Rev Immunol*. 2014;14(4):247-259. doi:10.1038/nri3632
- Kulig M, Bergmann R, Klettke U, Wahn V, Tacke U, Wahn U. Natural course of sensitization to food and inhalant allergens during the first 6 years of life. *Journal of Allergy and Clinical Immunology*. 1999;103(6):1173-1179. doi:10.1016/S0091-6749(99)70195-8
- 9. JH S, EC M, JM S, RA W. The natural history of egg allergy. *J Allergy Clin Immunol*. 2007;120(6):1413-1417. doi:10.1016/J.JACI.2007.09.040
- Savage J, Johns CB. Food Allergy: Epidemiology and Natural History. *Immunol Allergy Clin North Am.* Published online 2015. doi:10.1016/j.iac.2014.09.004

- Yu JW, Kagan R, Verreault N, et al. Accidental ingestions in children with peanut allergy. *Journal of Allergy and Clinical Immunology*. 2006;118(2):466-472. doi:10.1016/J.JACI.2006.04.024
- 12. Leung M, Clarke AE, Gabrielli S, et al. Risk of peanut- and tree-nutinduced anaphylaxis during Halloween, Easter and other cultural holidays in Canadian children. *CMAJ*. 2020;192(38):E1084-E1092. doi:10.1503/CMAJ.200034
- Antolín-Amérigo D, Manso L, Caminati M, et al. Quality of life in patients with food allergy. *Clin Mol Allergy*. 2016;14:4. doi:10.1186/s12948-016-0041-4
- Cummings AJ, Knibb RC, King RM, Lucas JS. The psychosocial impact of food allergy and food hypersensitivity in children, adolescents and their families: A review. *Allergy: European Journal of Allergy and Clinical Immunology*. 2010;65(8):933-945. doi:10.1111/J.1398-9995.2010.02342.X
- Avery NJ, King RM, Knight S, Hourihane JOB. Assessment of quality of life in children with peanut allergy. *Pediatric Allergy and Immunology*. 2003;14(5):378-382. doi:10.1034/J.1399-3038.2003.00072.X
- Sicherer SH, Noone SA, Muñoz-Furlong A. The impact of childhood food allergy on quality of life. *Annals of Allergy, Asthma & Immunology*. 2001;87(6):461-464. doi:10.1016/S1081-1206(10)62258-2
- 17. Gupta R, Holdford D, Bilaver L, Dyer A, Holl JL, Meltzer D. The Economic Impact of Childhood Food Allergy in the United States. *JAMA Pediatr*. 2013;167(11):1026-1031. doi:10.1001/JAMAPEDIATRICS.2013.2376
- 18. Waldmann TA, Iio A, Ogawa M, McIntyre OR, Strober W. The Metabolism of IgE. *The Journal of Immunology*. 1976;117(4).
- Jiménez-Saiz R, Chu DK, Mandur TS, et al. Lifelong memory responses perpetuate humoral T H 2 immunity and anaphylaxis in food allergy. *Journal of Allergy and Clinical Immunology*. 2017;140(6):1604-1615.e5. doi:10.1016/j.jaci.2017.01.018
- 20. P V, B P, G L. Platelet-activating factor, histamine, and tryptase levels in human anaphylaxis. *J Allergy Clin Immunol*. 2013;131(1):144-149. doi:10.1016/J.JACI.2012.08.016
- 21. Arias K, Baig M, Colangelo M, et al. Concurrent blockade of plateletactivating factor and histamine prevents life-threatening peanut-induced

anaphylactic reactions. *Journal of Allergy and Clinical Immunology*. 2009;124(2):307-314.e2. doi:10.1016/J.JACI.2009.03.012

- 22. Reber LL, Hernandez JD, Galli SJ. The pathophysiology of anaphylaxis. *J Allergy Clin Immunol.* 2017;140(2):335. doi:10.1016/J.JACI.2017.06.003
- Charlesworth EN, Hood AF, Soter NA, Kagey-Sobotka A, Norman PS, Lichtenstein LM. Cutaneous late-phase response to allergen. Mediator release and inflammatory cell infiltration. *Journal of Clinical Investigation*. 1989;83(5):1519. doi:10.1172/JCI114047
- 24. Monti G, Muratore MC, Peltran A, et al. High incidence of adverse reactions to egg challenge on first known exposure in young atopic dermatitis children: predictive value of skin prick test and radioallergosorbent test to egg proteins. *Clinical & Experimental Allergy*. 2002;32(10):1515-1519. doi:10.1046/J.1365-2745.2002.01454.X
- 25. Azad MB, Konya T, Guttman DS, et al. Infant gut microbiota and food sensitization: associations in the first year of life. *Clinical & Experimental Allergy*. 2015;45(3):632-643. doi:10.1111/CEA.12487
- 26. Du Toit G, Roberts G, Sayre PH, et al. Randomized Trial of Peanut Consumption in Infants at Risk for Peanut Allergy. http://dx.doi.org/101056/NEJMoa1414850. 2015;372(9):803-813. doi:10.1056/NEJMOA1414850
- 27. G L, D F, K N, J G. Factors associated with the development of peanut allergy in childhood. *N Engl J Med.* 2003;348(11):977-985. doi:10.1056/NEJMOA013536
- SJ B, Y A, HJ C, et al. Loss-of-function variants in the filaggrin gene are a significant risk factor for peanut allergy. *J Allergy Clin Immunol*. 2011;127(3):661-667. doi:10.1016/J.JACI.2011.01.031
- 29. SE A, HT T, P V, et al. The skin barrier function gene SPINK5 is associated with challenge-proven IgE-mediated food allergy in infants. *Allergy*. 2017;72(9):1356-1364. doi:10.1111/ALL.13143
- 30. HA B, AH L, S, et al. Atopic dermatitis increases the effect of exposure to peanut antigen in dust on peanut sensitization and likely peanut allergy. J Allergy Clin Immunol. 2015;135(1):164-170.e4. doi:10.1016/J.JACI.2014.10.007

- Amanna IJ, Carlson NE, Slifka MK. Duration of humoral immunity to common viral and vaccine antigens. *New England Journal of Medicine*. Published online 2007. doi:10.1056/NEJMoa066092
- 32. Asrat S, Kaur N, Liu X, et al. Chronic allergen exposure drives accumulation of long-lived IgE plasma cells in the bone marrow, giving rise to serological memory. *Sci Immunol*. 2020;5(43):8402. doi:10.1126/sciimmunol.aav8402
- 33. Beeh KM, Beier J, Buhl R. Seasonal variations of serum-IgE and potential impact on dose-calculation of omalizumab (rhuMab-E25, anti-IgE). *Pneumologie*. Published online 2004. doi:10.1055/s-2004-818483
- 34. Carballeda-Sangiao N, Rodríguez-Mahillo AI, Careche M, Navas A, Moneo I, González-Muñoz M. Changes over Time in IgE Sensitization to Allergens of the Fish Parasite Anisakis spp. *PLoS Negl Trop Dis*. Published online 2016. doi:10.1371/journal.pntd.0004864
- 35. Kim MS, Straesser MD, Keshavarz B, et al. IgE to galactose-α-1,3galactose wanes over time in patients who avoid tick bites. *J Allergy Clin Immunol Pract*. 2020;8(1):364-367.e2. doi:10.1016/J.JAIP.2019.08.045
- Commins SP, James HR, Kelly LA, et al. The relevance of tick bites to the production of IgE antibodies to the mammalian oligosaccharide galactoseα-1,3-galactose. *J Allergy Clin Immunol*. 2011;127(5):1286. doi:10.1016/J.JACI.2011.02.019
- 37. Fearon DT, Manders P, Wagner SD. Arrested differentiation, the selfrenewing memory lymphocyte, vaccination. *Science (1979)*. Published online 2001. doi:10.1126/science.1062589
- Yang Z, Robinson MJ, Allen CDC. Regulatory constraints in the generation and differentiation of IgE-expressing B cells. *Curr Opin Immunol.* Published online 2014. doi:10.1016/j.coi.2014.02.001
- Jiménez-Saiz R, Ellenbogen Y, Bruton K, et al. Human BCR analysis of single-sorted, putative IgE+ memory B cells in food allergy. *Journal of Allergy and Clinical Immunology*. Published online 2019. doi:10.1016/j.jaci.2019.04.001
- 40. Looney TJ, Lee JY, Roskin KM, et al. Human B-cell isotype switching origins of IgE. *Journal of Allergy and Clinical Immunology*. Published online 2016. doi:10.1016/j.jaci.2015.07.014

- 41. Jiménez-Saiz R, Ellenbogen Y, Koenig JFE, et al. IgG1 + B-cell immunity predates IgE responses in epicutaneous sensitization to foods. *Allergy: European Journal of Allergy and Clinical Immunology*. Published online 2019. doi:10.1111/all.13481
- 42. Xiong H, Dolpady J, Wabl M, et al. Sequential class switching is required for the generation of high affinity IgE antibodies. *Journal of Experimental Medicine*. 2012;209(2):353-364. doi:10.1084/jem.20111941
- 43. He JS, Subramaniam S, Narang V, et al. IgG1 memory B cells keep the memory of IgE responses. *Nat Commun.* 2017;8(1). doi:10.1038/s41467-017-00723-0
- 44. Turqueti-Neves A, Otte M, Schwartz C, et al. The Extracellular Domains of IgG1 and T Cell-Derived IL-4/IL-13 Are Critical for the Polyclonal Memory IgE Response In Vivo. *PLoS Biol*. Published online 2015. doi:10.1371/journal.pbio.1002290
- Moutsoglou DM, Dreskin SC. B cells establish, but do not maintain, longlived murine anti-peanut IgEa. *Clinical and Experimental Allergy*. 2016;46(4):640-653. doi:10.1111/cea.12715
- 46. Harris DP, Goodrich S, Mohrs K, Mohrs M, Lund FE. Cutting Edge: The Development of IL-4-Producing B Cells (B Effector 2 Cells) Is Controlled by IL-4, IL-4 Receptor α, and Th2 Cells. *The Journal of Immunology*. 2005;175(11):7103-7107. doi:10.4049/JIMMUNOL.175.11.7103
- 47. Finkelman FD, Katona IM, Urban JF, et al. IL-4 is required to generate and sustain in vivo IgE responses. *The Journal of Immunology*. 1988;141(7).
- IL K, M M. IL-4-producing CD4+ T cells in reactive lymph nodes during helminth infection are T follicular helper cells. *J Exp Med*. 2009;206(5):1001-1007. doi:10.1084/JEM.20090313
- 49. Mohrs K, Wakil AE, Killeen N, Locksley RM, Mohrs M. A two-step process for cytokine production revealed by IL-4 dual-reporter mice. *Immunity*. Published online 2005. doi:10.1016/j.immuni.2005.09.006
- 50. Bao K, Reinhardt RL. The differential expression of IL-4 and IL-13 and its impact on type-2 immunity. *Cytokine*. 2015;75(1):25-37.
- 51. Liang HE, Reinhardt RL, Bando JK, Sullivan BM, Ho IC, Locksley RM. Divergent expression patterns of IL-4 and IL-13 define unique functions in allergic immunity. *Nature Immunology 2011 13:1*. 2011;13(1):58-66. doi:10.1038/ni.2182

- 52. Grünig G, Warnock M, Wakil AE, et al. Requirement for IL-13 independently of IL-4 in experimental asthma. *Science (1979)*. Published online 1998. doi:10.1126/science.282.5397.2261
- 53. Bitton A, Avlas S, Reichman H, et al. A key role for IL-13 signaling via the type 2 IL-4 receptor in experimental atopic dermatitis. *Sci Immunol*. Published online 2020. doi:10.1126/sciimmunol.aaw2938
- 54. Castro M, Corren J, Pavord ID, et al. Dupilumab Efficacy and Safety in Moderate-to-Severe Uncontrolled Asthma. *New England Journal of Medicine*. Published online 2018. doi:10.1056/NEJMoa1804092
- 55. Beck LA, Thaçi D, Hamilton JD, et al. Dupilumab treatment in adults with moderate-to-severe atopic dermatitis. *New England Journal of Medicine*. Published online 2014. doi:10.1056/NEJMoa1314768
- 56. Bruton K, Spill P, Vohra S, et al. Interrupting reactivation of immunologic memory diverts the allergic response and prevents anaphylaxis. *Journal of Allergy and Clinical Immunology*. 2021;147(4):1381-1392. doi:10.1016/J.JACI.2020.11.042
- 57. Neuman-Sunshine DL, Eckman JA, Keet CA, et al. The natural history of persistent peanut allergy. *Annals of Allergy, Asthma and Immunology*. Published online 2012. doi:10.1016/j.anai.2011.11.010
- 58. Jones SM, Burks AW, Dupont C. State of the art on food allergen immunotherapy: Oral, sublingual, and epicutaneous. *Journal of Allergy and Clinical Immunology*. Published online 2014. doi:10.1016/j.jaci.2013.12.1040
- 59. Durham SR, Penagos M. Sublingual or subcutaneous immunotherapy for allergic rhinitis? *Journal of Allergy and Clinical Immunology*. Published online 2016. doi:10.1016/j.jaci.2015.12.1298
- 60. Calderon MA, Alves B, Jacobson M, Hurwitz B, Sheikh A, Durham S. Allergen injection immunotherapy for seasonal allergic rhinitis. *Cochrane Database of Systematic Reviews*. Published online 2007. doi:10.1002/14651858.CD001936.pub2
- 61. Durham SR, Walker SM, Varga EM, et al. Long-Term Clinical Efficacy of Grass-Pollen Immunotherapy. http://dx.doi.org/101056/NEJM199908123410702. 2008;341(7):468-475. doi:10.1056/NEJM199908123410702

- 62. Naclerio RM, Proud D, Moylan B, et al. Clinical aspects of allergic disease: A double-blind study of the discontinuation of ragweed immunotherapy. *Journal of Allergy and Clinical Immunology*. 1997;100(3):293-300. doi:10.1016/S0091-6749(97)70240-9
- 63. Walker SM, Varney VA, Gaga M, Jacobson MR, Durham SR. dag. *Allergy*. 1995;50(5):405-413. doi:10.1111/J.1398-9995.1995.TB01170.X
- 64. Corren J, Saini SS, Gagnon R, et al. Short-Term Subcutaneous Allergy Immunotherapy and Dupilumab are Well Tolerated in Allergic Rhinitis: A Randomized Trial. *J Asthma Allergy*. 2021;14:1045. doi:10.2147/JAA.S318892
- 65. Oppenheimer JJ, Nelson HS, Bock SA, Christensen F, Leung DYM. Treatment of peanut allergy with rush immunotherapy. *J Allergy Clin Immunol*. Published online 1992. doi:10.1016/0091-6749(92)90080-L
- 66. HS N, J L, R R, A B, D L. Treatment of anaphylactic sensitivity to peanuts by immunotherapy with injections of aqueous peanut extract. *J Allergy Clin Immunol*. 1997;99(6 Pt 1):744-751. doi:10.1016/S0091-6749(97)80006-1
- 67. Kim EH, Burks AW. Food allergy immunotherapy: Oral immunotherapy and epicutaneous immunotherapy. *Allergy*. 2020;75(6):1337-1346. doi:10.1111/ALL.14220
- 68. G S, N G, S H, et al. Epicutaneous allergen administration as a novel method of allergen-specific immunotherapy. *J Allergy Clin Immunol*. 2009;124(5):997-1002. doi:10.1016/J.JACI.2009.07.019
- 69. Dupont C, Kalach N, Soulaines P, Legoué-Morillon S, Piloquet H, Benhamou PH. *Cow's Milk Epicutaneous Immunotherapy in Children: A Pilot Trial of Safety, Acceptability, and Impact on Allergic Reactivity.* Vol 125. Elsevier; 2010:1165-1167. doi:10.1016/j.jaci.2010.02.029
- 70. Jones SM, Sicherer SH, Burks AW, et al. Epicutaneous immunotherapy for the treatment of peanut allergy in children and young adults. *Journal of Allergy and Clinical Immunology*. Published online 2017. doi:10.1016/j.jaci.2016.08.017
- 71. Sampson HA, Shreffler WG, Yang WH, et al. Effect of Varying Doses of Epicutaneous Immunotherapy vs Placebo on Reaction to Peanut Protein Exposure Among Patients With Peanut Sensitivity: A Randomized Clinical Trial. JAMA. 2017;318(18):1798-1809. doi:10.1001/JAMA.2017.16591

- 72. Fleischer DM, Greenhawt M, Sussman G, et al. Effect of Epicutaneous Immunotherapy vs Placebo on Reaction to Peanut Protein Ingestion among Children with Peanut Allergy: The PEPITES Randomized Clinical Trial. In: JAMA - Journal of the American Medical Association. ; 2019. doi:10.1001/jama.2019.1113
- 73. Burks AW, Sampson HA, Plaut M, Lack G, Akdis CA. Treatment for food allergy. *Journal of Allergy and Clinical Immunology*. 2018;141(1):1-9. doi:10.1016/J.JACI.2017.11.004
- 74. Burks AW, Jones SM, Wood RA, et al. Oral immunotherapy for treatment of egg allergy in children. *New England Journal of Medicine*. Published online 2012. doi:10.1056/NEJMoa1200435
- Jones SM, Burks AW, Keet C, et al. Long-term treatment with egg oral immunotherapy enhances sustained unresponsiveness that persists after cessation of therapy. *Journal of Allergy and Clinical Immunology*. 2016;137(4):1117-1127.e10. doi:10.1016/J.JACI.2015.12.1316/ATTACHMENT/E703E32F-D109-44A7-83A6-754DD77C4C72/MMC1.PDF
- 76. Meglio P, Bartone E, Plantamura M, Arabito E, Giampietro PG. A protocol for oral desensitization in children with IgE-mediated cow's milk allergy. *Allergy*. 2004;59(9):980-987. doi:10.1111/J.1398-9995.2004.00542.X
- Pajno GB, Caminiti L, Ruggeri P, et al. Oral immunotherapy for cow's milk allergy with a weekly up-dosing regimen: a randomized single-blind controlled study. *Annals of Allergy, Asthma & Immunology*. 2010;105(5):376-381. doi:10.1016/J.ANAI.2010.03.015
- 78. Martorell A, De la Hoz B, Ibáñez MD, et al. Oral desensitization as a useful treatment in 2-year-old children with cow's milk allergy. *Clin Exp Allergy*. 2011;41(9):1297-1304. doi:10.1111/J.1365-2222.2011.03749.X
- 79. Wood RA, Kim JS, Lindblad R, et al. A randomized, double-blind, placebo-controlled study of omalizumab combined with oral immunotherapy for the treatment of cow's milk allergy. *Journal of Allergy and Clinical Immunology*. 2016;137(4):1103-1110.e11. doi:10.1016/J.JACI.2015.10.005
- Jones SM, Pons L, Roberts JL, et al. Clinical efficacy and immune regulation with peanut oral immunotherapy. *Journal of Allergy and Clinical Immunology*. Published online 2009. doi:10.1016/j.jaci.2009.05.022

- Varshney P, Jones SM, Scurlock AM, et al. A randomized controlled study of peanut oral immunotherapy: Clinical desensitization and modulation of the allergic response. *Journal of Allergy and Clinical Immunology*. 2011;127(3):654-660. doi:10.1016/J.JACI.2010.12.1111
- 82. Vickery BP, Scurlock AM, Kulis M, et al. Sustained unresponsiveness to peanut in subjects who have completed peanut oral immunotherapy. *Journal of Allergy and Clinical Immunology*. Published online 2014. doi:10.1016/j.jaci.2013.11.007
- 83. Chinthrajah RS, Purington N, Andorf S, et al. Sustained outcomes in oral immunotherapy for peanut allergy (POISED study): a large, randomised, double-blind, placebo-controlled, phase 2 study. *The Lancet*. Published online 2019. doi:10.1016/S0140-6736(19)31793-3
- Chu DK, Wood RA, French S, et al. Oral immunotherapy for peanut allergy (PACE): a systematic review and meta-analysis of efficacy and safety. *The Lancet*. Published online 2019. doi:10.1016/S0140-6736(19)30420-9
- 85. Yee CSK, Albuhairi S, Noh E, et al. Long-Term Outcome of Peanut Oral Immunotherapy Facilitated Initially by Omalizumab. *Journal of Allergy and Clinical Immunology: In Practice*. Published online 2019. doi:10.1016/j.jaip.2018.09.015
- Varshney P, Steele PH, Vickery BP, et al. Adverse reactions during peanut oral immunotherapy home dosing. *Journal of Allergy and Clinical Immunology*. 2009;124(6):1351-1352. doi:10.1016/J.JACI.2009.09.042
- Staden U, Rolinck-Werninghaus C, Brewe F, Wahn U, Niggemann B, Beyer K. Specific oral tolerance induction in food allergy in children: efficacy and clinical patterns of reaction. *Allergy*. 2007;62(11):1261-1269. doi:10.1111/J.1398-9995.2007.01501.X
- Schneider LC, Rachid R, Lebovidge J, Blood E, Mittal M, Umetsu DT. A pilot study of omalizumab to facilitate rapid oral desensitization in high-risk peanut-allergic patients. *Journal of Allergy and Clinical Immunology*. Published online 2013. doi:10.1016/j.jaci.2013.09.046
- 89. MacGinnitie AJ, Rachid R, Gragg H, et al. Omalizumab facilitates rapid oral desensitization for peanut allergy. *Journal of Allergy and Clinical Immunology*. 2017;139(3):873-881.e8. doi:10.1016/J.JACI.2016.08.010
- 90. Brandström J, Vetander M, Sundqvist AC, et al. Individually dosed omalizumab facilitates peanut oral immunotherapy in peanut allergic

adolescents. *Clinical & Experimental Allergy*. 2019;49(10):1328-1341. doi:10.1111/CEA.13469

- 91. Yee CSK, Albuhairi S, Noh E, et al. Long-Term Outcome of Peanut Oral Immunotherapy Facilitated Initially by Omalizumab. *J Allergy Clin Immunol Pract*. 2019;7(2):451-461.e7. doi:10.1016/J.JAIP.2018.09.015
- 92. Langlois A, Lavergne MH, Leroux H, et al. Protocol for a double-blind, randomized controlled trial on the dose-related efficacy of omalizumab in multi-food oral immunotherapy. *Allergy, Asthma and Clinical Immunology*. 2020;16(1):1-16. doi:10.1186/S13223-020-00419-Z/FIGURES/4
- 93. Cox LS, Linnemann DL, Nolte H, Weldon D, Finegold I, Nelson HS. Sublingual immunotherapy: A comprehensive review. *Journal of Allergy* and Clinical Immunology. 2006;117(5):1021-1035. doi:10.1016/J.JACI.2006.02.040/ATTACHMENT/554C67C7-7C9B-4AFB-9646-7D9445036AE8/MMC1.PDF
- 94. Wilson DR, Lima MT, Durham SR. Sublingual immunotherapy for allergic rhinitis: systematic review and meta-analysis*. *Allergy*. 2005;60(1):4-12. doi:10.1111/J.1398-9995.2005.00699.X
- 95. Marogna M, Spadolini I, Massolo A, Canonica GW, Passalacqua G. Clinical, functional, and immunologic effects of sublingual immunotherapy in birch pollinosis: A 3-year randomized controlled study. *Journal of Allergy and Clinical Immunology*. 2005;115(6):1184-1188. doi:10.1016/J.JACI.2005.02.031
- 96. Marogna M, Spadolini I, Massolo A, Canonica GW, Passalacqua G. Randomized controlled open study of sublingual immunotherapy for respiratory allergy in real-life: clinical efficacy and more. *Allergy*. 2004;59(11):1205-1210. doi:10.1111/J.1398-9995.2004.00508.X
- 97. Dahl R, Kapp A, Colombo G, et al. Efficacy and safety of sublingual immunotherapy with grass allergen tablets for seasonal allergic rhinoconjunctivitis. *Journal of Allergy and Clinical Immunology*. Published online 2006. doi:10.1016/j.jaci.2006.05.003
- 98. Dahl R, Kapp A, Colombo G, et al. Sublingual grass allergen tablet immunotherapy provides sustained clinical benefit with progressive immunologic changes over 2 years. *Journal of Allergy and Clinical Immunology*. Published online 2008. doi:10.1016/j.jaci.2007.10.039
- 99. Novembre E, Cianferoni A, Capra L, et al. Severe anaphylaxis to kiwi fruit: Immunologic changes related to successful sublingual allergen

immunotherapy. *Journal of Allergy and Clinical Immunology*. 2003;111(6):1406-1409. doi:10.1067/MAI.2003.1497

- 100. Kerzl R, Simonowa A, Ring J, Ollert M, Mempel M. Life-threatening anaphylaxis to kiwi fruit: Protective sublingual allergen immunotherapy effect persists even after discontinuation. *Journal of Allergy and Clinical Immunology*. 2007;119(2):507-508. doi:10.1016/J.JACI.2006.09.041
- 101. Enrique E, Pineda F, Malek T, et al. Sublingual immunotherapy for hazelnut food allergy: A randomized, double-blind, placebo-controlled study with a standardized hazelnut extract. *Journal of Allergy and Clinical Immunology*. Published online 2005. doi:10.1016/j.jaci.2005.08.027
- 102. Fernández-Rivas M, Garrido Fernández S, Nadal JA, et al. Randomized double-blind, placebo-controlled trial of sublingual immunotherapy with a Pru p 3 quantified peach extract. *Allergy*. 2009;64(6):876-883. doi:10.1111/J.1398-9995.2008.01921.X
- 103. Rodriguez MJ, Mascaraque A, Ramos-Soriano J, et al. Pru p 3-Epitopebased sublingual immunotherapy in a murine model for the treatment of peach allergy. *Mol Nutr Food Res.* 2017;61(10). doi:10.1002/MNFR.201700110/FORMAT/PDF
- 104. Rodriguez MJ, Ramos-Soriano J, Perkins JR, et al. Glycosylated nanostructures in sublingual immunotherapy induce long-lasting tolerance in LTP allergy mouse model. *Scientific Reports 2019 9:1*. 2019;9(1):1-9. doi:10.1038/s41598-019-40114-7
- 105. De Boissieu D, Dupont C. Sublingual immunotherapy for cow's milk protein allergy: a preliminary report. *Allergy*. 2006;61(10):1238-1239. doi:10.1111/J.1398-9995.2006.01196.X
- 106. Keet CA, Frischmeyer-Guerrerio PA, Thyagarajan A, et al. The safety and efficacy of sublingual and oral immunotherapy for milk allergy. *Journal of Allergy and Clinical Immunology*. Published online 2012. doi:10.1016/j.jaci.2011.10.023
- 107. Fleischer DM, Burks AW, Vickery BP, et al. Sublingual immunotherapy for peanut allergy: A randomized, double-blind, placebo-controlled multicenter trial. *Journal of Allergy and Clinical Immunology*. Published online 2013. doi:10.1016/j.jaci.2012.11.011
- 108. Burks AW, Wood RA, Jones SM, et al. Sublingual immunotherapy for peanut allergy: Long-term follow-up of a randomized multicenter trial.

Journal of Allergy and Clinical Immunology. Published online 2015. doi:10.1016/j.jaci.2014.12.1917

- 109. Kim EH, Bird JA, Kulis M, et al. Sublingual immunotherapy for peanut allergy: Clinical and immunologic evidence of desensitization. *Journal of Allergy and Clinical Immunology*. Published online 2011. doi:10.1016/j.jaci.2010.12.1083
- 110. Kim EH, Yang L, Ye P, et al. Long-term sublingual immunotherapy for peanut allergy in children: Clinical and immunologic evidence of desensitization. *Journal of Allergy and Clinical Immunology*. Published online 2019. doi:10.1016/j.jaci.2019.07.030
- 111. Narisety SD, Frischmeyer-Guerrerio PA, Keet CA, et al. A randomized, double-blind, placebo-controlled pilot study of sublingual versus oral immunotherapy for the treatment of peanut allergy. *Journal of Allergy and Clinical Immunology*. Published online 2015. doi:10.1016/j.jaci.2014.11.005
- 112. Orsini Delgado ML, Rizzo GP, Fossati CA, et al. Sublingual Omp16driven redirection of the allergic intestinal response in a pre-clinical model of food allergy. *Clin Exp Allergy*. 2020;50(8):954-963. doi:10.1111/CEA.13676
- 113. Sun J, Arias K, Alvarez D, et al. Impact of CD40 Ligand, B Cells, and Mast Cells in Peanut-Induced Anaphylactic Responses. *The Journal of Immunology*. 2007;179(10):6696-6703. doi:10.4049/jimmunol.179.10.6696
- 114. Kong J, Chalcraft K, Mandur TS, et al. Comprehensive metabolomics identifies the alarmin uric acid as a critical signal for the induction of peanut allergy. *Allergy: European Journal of Allergy and Clinical Immunology*. Published online 2015. doi:10.1111/all.12579
- 115. Arias K, Chu DK, Flader K, et al. Distinct immune effector pathways contribute to the full expression of peanut-induced anaphylactic reactions in mice. *Journal of Allergy and Clinical Immunology*. 2011;127(6):1552-1561.e1. doi:10.1016/J.JACI.2011.03.044
- 116. Zhang B, Liu E, Gertie JA, et al. Divergent T follicular helper cell requirement for IgA and IgE production to peanut during allergic sensitization. *Sci Immunol*. Published online 2020. doi:10.1126/sciimmunol.aay2754
- 117. Gagliardi MC, Sallusto F, Marinaro M, Langenkamp A, Lanzavecchia A, de Magistris MT. Cholera toxin induces maturation of human dendritic

cells and licences them for Th2 priming. *Wiley Online Library*. Accessed September 11, 2022. https://onlinelibrary.wiley.com/doi/abs/10.1002/1521-4141(2000)30:8%3C2394::AID-IMMU2394%3E3.0.CO;2-Y

- Richman LK, Chiller JM, Brown WR, Hanson DG, Vaz NM. Enterically Induced Immunologic Tolerance. *The Journal of Immunology*. 1978;121(6).
- 119. Ngan J, Kind LS. Suppressor T Cells for IgE and IgG in Peyer's Patches of Mice Made Tolerant by the Oral Administration of Ovalbumin. *The Journal of Immunology*. 1978;120(3).
- 120. Miller SD, Hanson DG. Inhibition of Specific Immune Responses by Feeding Protein Antigens. *The Journal of Immunology*. 1979;123(5).
- 121. Lundsgaard EA, Danneskiold-Samsøe NB, Sonne SB, et al. Preventive sublingual immunotherapy with peanut allergen ara h 6 drives expansion of helios regulatory T cells and alters expression of genes involved in maintenance of sublingual tissue integrity. In: *European Academy of Allergy and Clinical Immunology Congress 2021*. ; 2021.
- 122. Burton OT, Tamayo JM, Stranks AJ, Koleoglou KJ, Oettgen HC. Allergenspecific IgG antibody signaling through FcγRIIb promotes food tolerance. *Journal of Allergy and Clinical Immunology*. 2018;141(1):189-201.e3. doi:10.1016/j.jaci.2017.03.045
- 123. Finkelman FD, Rothenberg ME, Brandt EB, Morris SC, Strait RT. Molecular mechanisms of anaphylaxis: Lessons from studies with murine models. *Journal of Allergy and Clinical Immunology*. 2005;115(3):449-457. doi:10.1016/J.JACI.2004.12.1125
- 124. Catalán D, Mansilla MA, Ferrier A, et al. Immunosuppressive Mechanisms of Regulatory B Cells. *Front Immunol*. 2021;12:611795. doi:10.3389/FIMMU.2021.611795
- 125. Ma S, Satitsuksanoa P, Jansen K, Cevhertas L, van de Veen W, Akdis M. B regulatory cells in allergy. *Immunol Rev.* 2021;299(1):10-30. doi:10.1111/IMR.12937
- 126. Smaldini PL, Trejo F, Cohen JL, Piaggio E, Docena GH. Systemic IL-2/anti-IL-2Ab complex combined with sublingual immunotherapy suppresses experimental food allergy in mice through induction of mucosal regulatory T cells. *Allergy*. 2018;73(4):885-895. doi:10.1111/ALL.13402

127. Lucía M, Delgado O, Gastón |, et al. Sublingual Omp16-driven redirection of the allergic intestinal response in a pre-clinical model of food allergy. *Clinical & Experimental Allergy*. 2020;50(8):954-963. doi:10.1111/CEA.13676